

Delft University of Technology

A system dynamics model for analyzing modal shift policies towards decarbonization in freight transportation

Nassar, Raphael Ferrari; Ghisolfi, Verônica; Annema, Jan Anne; van Binsbergen, Arjan; Tavasszy, Lóránt Antal

DOI 10.1016/j.rtbm.2023.100966

Publication date 2023 Document Version Final published version Published in Research in Transportation Business and Management

Citation (APA)

Nassar, R. F., Ghisolfi, V., Annema, J. A., van Binsbergen, A., & Tavasszy, L. A. (2023). A system dynamics model for analyzing modal shift policies towards decarbonization in freight transportation. *Research in Transportation Business and Management*, *48*, Article 100966. https://doi.org/10.1016/j.rtbm.2023.100966

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect



Research in Transportation Business & Management

journal homepage: www.elsevier.com/locate/rtbm



A system dynamics model for analyzing modal shift policies towards decarbonization in freight transportation

Raphael Ferrari Nassar^{a,b,*}, Verônica Ghisolfi^c, Jan Anne Annema^a, Arjan van Binsbergen^b, Lóránt Antal Tavasszy^{a,b}

^a Delft University of Technology, Faculty of Technology, Policy and Management, the Netherlands

^b Delft University of Technology, Faculty of Civil Engineering Geosciences, the Netherlands

^c Transportation Engineering Programme, Federal University of Rio de Janeiro—UFRJ, Brazil

ARTICLE INFO

Keywords: System dynamics Modal shift Freight transportation Decarbonization Policy analysis

ABSTRACT

The decarbonization of freight transport is currently a big challenge to tackle. One way of decarbonizing the sector is the modal shift towards the least polluting transport modes. This research aims to shed light on modal shift time dynamics in developing economies and considers whether a System Dynamics approach can assist with the policy-making decision about a modal shift towards freight decarbonization. This research explores policies that promote the modal shift of freight transportation for a Brazilian case study, using a System Dynamics model. Policies include fiscal and regulatory measures and infrastructure investments. The findings show that the process of modal shift is slow. However, implementing a combination of stricter policy measures early on, and changes in infrastructure investment strategies, accelerate the shift and this seems to be a robust measure package, capable of promoting a modal shift measures, which tend to lose efficiency over time, slowing down the pace of decarbonization. Findings also display how modal shift policies alone might not be sufficient to achieve a reduction in CO_2 emission. Addressing the problem with a System Dynamics approach may help decision-makers in economically developing countries to develop more effective policy strategies.

1. Introduction

The transport of goods is essential for economic development. However, freight transport also plays a significant role when it comes to resource consumption, pollution, and climate change. In 2017, road freight transportation accounted for around 7% of the total world energy-related carbon dioxide (CO₂) emissions (Kaack, Vaishnav, Morgan, Azevedo, & Rai, 2018). Freight demand is expected to increase and as this sector relies heavily on fossil fuels, decarbonization in freight transport seems to be a difficult challenge to be tackled.

Shifting freight transportation from road to rail would entail a reduction of CO_2 emissions (Hou & Geerlings, 2016). Nevertheless, what most countries have experienced is a growth in road freight and a shift from rail to road (Kaack et al., 2018). The modal shift can be promoted by having policies targeting certain infrastructure investments (e.g. new infrastructure projects or by improving existing infrastructure), fiscal measures (e.g. fuel pricing or road pricing), and regulatory measures (e.

g. truck size and weight limits) (Bickford et al., 2014; Woodburn, Browne, Piotrowska, & Allen, 2007).

Political decisions on new infrastructure projects, as well as fiscal and regulatory measures to promote modal shift, are long-term commitments and involve multiple stakeholders. The implementation of such decisions is time demanding and the impact cannot be measured right away, creating a complex and time-dependent, dynamic problem. Moreover, the time window to apply decarbonization policies is relatively short as many of the political goals to achieve CO_2 emission reductions are expected to be achieved in the coming decades (European Commission, 2011; European Environment Agency, 2019).

Kaack et al. (2018) mentioned the importance of looking into emerging economies and developing countries when looking for ways to replace road transport with rail transportation. Developing countries experience rapid growth in transportation demand, which entails the need for new transportation infrastructure. Brazil's economy relies heavily on exports and the country has an uneven structure when it

* Corresponding author at: Delft University of Technology, Faculty of Technology, Policy and Management, Faculty of Civil Engineering Geosciences, 2628 CN Delft, the Netherlands.

E-mail address: raphaelfnassar@gmail.com (R.F. Nassar).

https://doi.org/10.1016/j.rtbm.2023.100966

Received 31 December 2021; Received in revised form 16 February 2023; Accepted 6 March 2023 Available online 5 April 2023

2210-5395/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

comes to freight transportation, its current modal split consists of 65% via roadways, 15% via railways, and 20% via waterways (ANTF - Associação Nacional dos Transportadores Ferroviários, 2020). The demand for new infrastructure in Brazil is peaking, the country is under an intense private-public partnership (PPP) program as a strategy to overcome the limitations of its infrastructure network, focusing on improving its rail and waterways corridors, which makes Brazil an interesting country to study freight modal shift policies.

To analyze the modal shift impacts in such a system, it is necessary to look beyond isolated events and measures and study the interaction of different parts of the system and how these parts relate through time. System Dynamics (SD) is an approach that allows the study of the impact of different policies in such complex systems (Pruyt, 2013). The dynamics' complexity effect arises from the freight system's multiple feedback loops due to many interconnected components. This research aims to design and apply an SD model that can estimate future CO₂ emission impacts of modal shift policy measures over time, in the Brazilian context. We incorporated infrastructure projects, pricing measures, and stakeholders' decisions in our application to allow decisionmakers to understand the complexity of the Brazilian freight system and how it will develop its CO₂ footprint over time using different possible modal shift policy scenarios. Our scientific contribution is that, as far as we know, no study has analyzed the usefulness of a detailed SD model to determine which policy measures could be considered in Brazil to accelerate the modal shift, as well as the lag time the Brazilian freight system needs to adapt to the policies. By using SD, we can estimate the adaption times that are required in the different modal shift policies before CO₂ emission takes off. In the discussion, we also address the extent to which our findings can be generalized for other emerging economies.

We conducted a literature review on SD methodology applications in freight systems to define the basic setup of our model. From the literature, we extracted the elements needed to simulate the freight modal shift process at a national level. In the next step, we conducted interviews with experts, trying to grasp important aspects and factors for our model that were not fully covered in the literature. Using the findings from the literature and the interviews, we developed a causal loop diagram (CLD), illustrating the key relations in the system, followed by the development of the stock and flow diagram. We verified and validated the model using tests as proposed in the SD literature. Finally, we applied the SD model to analyze the behavior of the system when implementing policy measures to study the promotion of a modal shift and the reduction of CO_2 emissions.

The remainder of the paper is organized as follows: Section 2 describes the relevant literature on System Dynamics and its application to transportation problems, focusing on promoting a modal shift towards lower emission mode. Section 3 summarizes the process of the interview with experts. Section 4 presents the SD model developed for this research. Section 5 briefly introduces the validation and verification tests that were performed to ensure that the model was implemented following its specification and that it was an accurate representation of the system. Section 6 presents the experiments carried out and the discussion of the results. Lastly, Section 7 shows the conclusions and recommendations of this research.

2. Literature on freight decarbonization SD modeling: positioning our SD modeling choice

In this section, the state of the art in SD modeling in the research fields of freight transport, mode choice, and modal shift is shown. The approaches presented are investigated in detail and consideration is given regarding whether they are suitable for our research focus. Additionally, we position our choice for SD modeling within the wider freight modeling approaches.

The SD methodology was developed by Forrester (1961) as a basis of explanation to illustrate the effects of decisions in complex, dynamic

systems, in which the time functions are emphasized. The specific feature of SD is its non-linear feedback structures. For this reason, the interdependencies between system submodules have to be identified and illustrated in an iterative modeling procedure (Thaller, Clausen, & Kampmann, 2016). Abbas and Bell (1994) discussed and evaluated the strengths and weaknesses of SD concerning its suitability and appropriateness for transportation systems modeling. Shepherd (2014) presented a review of SD studies, categorizing them by area of application in the field of transportation and providing a summary of insights and recommendations for any future application of the SD approach in this field. There are a few studies regarding the mode choice or mode shift aspect.

Piattelli, Cuneo, Bianchi, and Soncin (2002) developed a macrolevel of variable aggregation approach, based on a generalized transportation model, to study the modal shift between roadways, railways, and waterways in Germany. They included three different public interventions in the model, possible investments in infrastructure for each mode, public coverage of operational costs by mode, and additional fuel taxation. Results showed that investments in infrastructure are more effective in promoting a decrease in freight transportation on roadways, compared to regulatory measures, such as fuel taxation. Han and Hayashi (2008) researched CO₂ mitigation scenarios in China's inter-city freight transport using SD modeling. They considered four different transport modes: railway, highway, waterway, and airway. Factors such as freight turnover volume, freight volume, network length, and fuel intensity were included in the model. The authors considered improving the traffic network (railway, highway, waterway, and airway growth rate) and fuel tax regulations as policy measures. Their simulation results showed that investing in the railway network is the most effective measure to reduce CO₂ emissions.

Liu, Mu, and Gong (2017) carried out a study using System Dynamics to evaluate modal shift policies to eliminate overload trucking in China, while promoting more sustainable alternatives such as railways. The authors considered rigid weight regulation and investments in railway infrastructure as policy measures. The results show that stricter weight regulations led to a higher total cumulative cost and did not have a positive impact on sustainability, while building a new railway could lead to an effective modal shift, removing freight from road and achieve better results in terms of sustainability. However, the research does not tackle the stakeholder aspect of the dynamics, and how they could react to certain changes in trend, e.g. if rail transport becomes financially more attractive than road, what would the trucks operators do to try to still keep their market share?

Choi, Park, and Lee (2019) developed a system dynamics model to analyze the impact of policy measures on promoting the modal shift from road to rail, using the steel industry for steel rolled coils transport in South Korea as the case study. In this study, containerization and taxes imposition on road transportation were applied as policy measures. The results show that containerization promoted the modal shift more rapidly than the taxations. The paper was able to provide a model that can support on anticipating the modal shift from road and to shed a light on promoting the modal shift using containerization. However, the rate to implement the policy measures was not provided and some aspects such as warehousing and transshipment costs were left out of the model.

GLADYSTE (Global Scale System Dynamic Simulation Model for Transport Emissions) is a model prototype that enabled the estimation of passenger and freight transport demand and emissions, and it simulates the impact of policy and technological measures in transport-related sectors, covering different transport modes and different regions of the world up to 2050 (Purwanto, González, Vanherle, Fermi, & Fiorello, 2011). The GLADYSTE model defined four main modules to simulate the transportation system and environmental impacts: Demand, Fleet, Environmental, and Welfare. However, the authors did not present any SD diagram and did not provide details in terms of lag time on the decisions' effects. Brito Junior et al. (2011) used an SD approach to study the modal shift of freight transport between road and waterways, as part of the decarbonization policy, using Brazil as a case study. They analyzed the modal shift based on the level of investment in the mode's capabilities and governmental pressure to reduce CO_2 emissions. They created a framework divided into four parts: transport capacity, transport demand, modal shift, and CO_2 emissions. The authors defined five different scenarios to simulate (pessimistic, moderate, moderate CO_2 , optimistic and optimistic CO_2) and compared the evolution of the modal shift between these two modes over fifteen years. Their results showed that the inertia for the maturation of the modal shift is long and stricter decarbonization policies could help accelerate the modal shift. However, despite the important results, the paper does not mention the stakeholders' involvement. It does not detail which measures could be considered to accelerate the modal shift, as well as the lag time to adapt to the new policies.

System dynamics is a tool that can also be used to analyze different type of transportation issues. While most of the papers we mentioned focused on inter-urban freight transportation, Astegiano, Fermi, and Martino (2019) built a modal to investigate the impact of electric bikes on modal split in Europe at an urban level and its effect on emission reduction. The simulation shows that e-bikes has a positive effect on reducing car and public transport modal share and contributes to make active modes more attractive.

Ghisolfi, Tavasszy, Correia, Chaves, and Ribeiro (2022) reviewed system dynamics models addressing different strategies for freight transport decarbonization, including the modal shift from road to alternative modes. They reviewed a total of 50 studies and highlighted the lack of transparency concerning how the time-dependent behavior is determined, especially on the time-leg and delay assumptions for each decision to achieve the results in some defined terms. In other words, these studies do not provide an understanding of how the system's interactions and processes can entail the time-lag needed for companies and the government to adapt and change to other transport modes as part of the decarbonization policy.

Ghisolfi et al. (2022) also shows how many of the SD papers focused on the decarbonization of freight transportation lack when it comes to presenting their model. Out of the 50 studies reviewed, only 19 presented both qualitative (causal loop diagram) and quantitative (Stock and Flow diagram) models, while 14 papers did not present any model. This creates a scientific gap as it makes it harder for new research to be built from what was done before. The paper brings to attention the lack of transparency when it comes to the source of some of the information used on these models and it touches on the lack of stakeholder involvement during the study, except for papers such as Seitz (2014).

Despite this gap, the brief overview presented above also shows how versatile this method can be. Through different research it was possible to tackle the same problem while considering different points, from modes contemplated to policy measures, using different assumptions, simulation timeframes, and countries as their case study, showing how wide the range of applicability such a tool has. What is especially unique in this study compared to other freight transport SD studies is that we expand the use of SD to tackle transportation problems by not only looking into new lag-time aspects of freight but also by evaluating different combinations of policy measures, intensity, and implementation time.

One of the advantages to use SD is also the lack of existing numerical data in the Brazilian context. Based on assumptions, interview results, and rough data, by using SD we can gain policy insights into what is an incomplete situation, data-wise. Other modelling approaches can use hypothetical data and rough assumptions, however, SD is considered adequate in structuring messy and wicked problems through 'group model building' approach (see Saryazdi, Ghatarim, Mashayekhi, & Hassanzadeh, 2020 for a large review on group model building). Saryazdi et al. (2020) shows that group model building is widely used to extract 'data' and insights from experts to build causal loop diagrams and quantitative stock and flow diagrams, also in the field of

transportation (i.e., Elias, Cavana, & Jackson, 2004).

Jonkeren, Francke, and Visser (2019) give an overview of methods for analyzing the modal shift in freight transport and its CO₂ gains. Discrete choice models, Life-Cycle approaches, simulation models (macro and micro), and even qualitative approaches are applied in this field. Based on their analysis, a possible alternative modeling route for us could have been discrete choice modeling in combination with some kind of network simulation modeling, for example. First, to understand the choices of Brazilian shippers (e.g. their mode choice) and the choices of Brazilian carriers (e.g. their route or vehicle choice), discrete choice modeling, in the context of new climate policies, could have been applied. Subsequently, the outcomes of these choice models could have flown in network simulation models, agent-based models, or even in the more classic four-stage transport models, among others, for analyzing potential feedback loops. For example, if the choice model shows that in the context of some modal shift policy more Brazilian shippers choose inland shipping but the waterway capacity lags behind, the simulation models would show longer transport times due to waterway congestion which in turn would influence the shippers' choices. Although combining these numerical modeling approaches would potentially be more accurate compared to our more assumption-based SD approach, the combination is also very time-consuming and data-intensive (and whether all the data required can be estimated with a sufficient amount of reliability is highly uncertain).

So, for this study, SD seems fit for purpose, but it should be noted that assumptions and rough insights play an important role in our SD outcomes, which means that the final results should be interpreted with some caution.

3. Interviews with experts

The literature review helped to build an initial understanding of the complexity of decarbonization policies applied to freight transport. However, there was still a need to understand some peculiarities not presented in scientific papers, such as the considerations made by infrastructure users regarding mode choice, how the different perspectives of the stakeholders involved in the policy-making process could impact the promotion of this modal shift, and how this could be presented time-wise.

We interviewed experts to better understand the dynamics between the actors involved in the freight transport policy-making process (specifically the carriers) and the governmental decisions about which projects to prioritize and invest in. We used an unstructured format (Figgou & Pavlopoulos, 2015), allowing experts to provide their opinion on the topic without the author's interference and thereby collecting data based on the expert's perspective and experiences (Nepveu, 2020). For this research, we selected six different Brazilians experts to talk to, from policy makers to port directors and exports/concessions experts, to have an overview of the topic. The experts interviewed can be seen in Table 1.

The experts provided insights on important variables included in the model and the "gaps" between decision-making and its effective implementation. Providing the necessary infrastructure on its own does not guarantee that a modal shift will take place; there is a need to dig deeper into the infrastructure user's perspective, comprehend what they consider necessary when choosing which infrastructure to use, know how they will include a modal shift into their strategic planning, and know how willing they are to shift from one mode to another.

In Table 2, a summary of all the time aspects/lag-time/delays questions we selected for our research is presented. We divided the table into two groups (1) Infrastructure Implementation and (2) Infrastructure Usage. Besides the identified time aspects/lag-time/delays, the table also presents which actors are involved in each one of them.

After the interviews, we organized the main findings into external and internal aspects of freight dynamics used as variables or as parameter values in the model. We highlight here the aspects related to the

Table 1

Interviewees overview.

Expert number	Role	Field of expertise
1	Director of Operations in a Brazilian port	Port operation
2	Project Manager working for the Brazilian government	Government strategic planning
3	Manager of Market Intelligence in a Brazilian railway company	Concession planning; new infrastructure projects; demand analysis
4	Executive Program Manager in a Brazilian railway company	New infrastructure planning; demand analysis
5	Senior Policy Specialist at a not-for-profit organization	Implementation of regulations; formulating the public policy for infrastructure projects
6	Logistics and Operations Manager in a commodity exports company	Export of agricultural commodities

Table 2

Summary of the time aspects/lag-time/delays.

GroupsIdentified time aspects/lag-time/delays How long does it take to elaborate the concession plan for a specific infrastructure project?Actors InvolvedHow long does it take to elaborate the concession plan for a specific infrastructure project?GovernmentHow long after the pre-project phase does it take to chose an infrastructure project to implement?GovernmentInfrastructure ImplementationWhat are the bureaucratic aspects involved in such projects and how can they delay their implementation?CarriersInfrastructure ImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project fund provided by the public sector?Government carriersInfrastructure UsageOnce the new infrastructure is available, how long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers Shippers			
concession plan for a specific infrastructure project?GovernmentHow long after the pre-project phase does it take to choose an infrastructure project to implement?GovernmentMat are the bureaucratic aspects involved in such projects and how can they delay their implementation?GovernmentInfrastructure ImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project fund provided by the public sector?Government CarriersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers	Groups		Actors Involved
How long after the pre-project phase does it take to choose an infrastructure project to implement?Government CarriersInfrastructure ImplementationShippersGovernmentInfrastructure ImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project fund provided by the public sector?Government CarriersHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersWhat are the aspects that might delay the CarriersCarriers Shippers		6	Government
does it take to chose an infrastructure project to implement?Carriers ShippersInfrastructureinvolved in such projects and how can involved in such projects and how can Uong does it normally take for the project to be implemented?Government CarriersImplementationOnce the company has been chosen, how Uong does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project fund provided by the public sector?Government CarriersHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersWhat are the aspects that might delay the CarriersCarriers		infrastructure project?	
project to implement?ShippersWhat are the bureaucratic aspectsGovernmentinvolved in such projects and how canCarriersInfrastructurethey delay their implementation?ShippersImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project fund provided by the publicGovernment carriersHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersWhat are the aspects that might delay theCarriers		How long after the pre-project phase	Government
What are the bureaucratic aspectsGovernmentInfrastructureinvolved in such projects and how canCarriersImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government CarriersHow long does it take to elaborate a project?Government CarriersHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Government CarriersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersWhat are the aspects that might delay theCarriers		does it take to choose an infrastructure	Carriers
Infrastructureinvolved in such projects and how can they delay their implementation?Carriers ShippersImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government GovernmentHow long does it take to elaborate a project fund provided by the public sector?Government GovernmentHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers Shippers		project to implement?	Shippers
Infrastructurethey delay their implementation?ShippersImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented?Government CarriersHow is delay considered when planning the execution of an infrastructure project?Government ProjectHow long does it take to elaborate a project fund provided by the public sector?Government GovernmentHow long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers ShippersInfrastructure UsageOnce the aspects that might delay the What are the aspects that might delay theCarriers Carriers		What are the bureaucratic aspects	Government
ImplementationOnce the company has been chosen, how long does it normally take for the project to be implemented? How is delay considered when planning the execution of an infrastructure project? How long does it take to elaborate a project fund provided by the public sector? How long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Government CarriersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay theCarriers		1 0	
Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the carriers in the orce of the other?			Shippers
to be implemented? How is delay considered when planning the execution of an infrastructure Government project? How long does it take to elaborate a project fund provided by the public Government sector? How long does it take to reach an equilibrium between carriers and Carriers shippers (supply and demand) on a new network? Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers	Implementation		Government
the execution of an infrastructure Government project? How long does it take to elaborate a project fund provided by the public Government sector? How long does it take to reach an equilibrium between carriers and Carriers shippers (supply and demand) on a new network? Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers		· · · ·	
project?How long does it take to elaborate aproject fund provided by the publicGovernmentsector?How long does it take to reach anequilibrium between carriers andCarriersshippers (supply and demand) on a newShippersnetwork?Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?CarriersWhat are the aspects that might delay theCarriers		How is delay considered when planning	
project fund provided by the public sector?Government sector?How long does it take to reach an equilibrium between carriers and shippers (supply and demand) on a new network?Carriers ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers ShippersWhat are the aspects that might delay theCarriers			Government
sector? How long does it take to reach an equilibrium between carriers and Carriers shippers (supply and demand) on a new Shippers network? Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers		How long does it take to elaborate a	
equilibrium between carriers and Carriers shippers (supply and demand) on a new Shippers network? Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers			Government
shippers (supply and demand) on a new network?ShippersInfrastructure UsageOnce the new infrastructure is available, how long does it take for shippers to shift from one mode to the other?Carriers ShippersWhat are the aspects that might delay theCarriers		How long does it take to reach an	
network? Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers		equilibrium between carriers and	Carriers
Infrastructure Usage Once the new infrastructure is available, how long does it take for shippers to shift from one mode to the other? Carriers What are the aspects that might delay the Carriers		shippers (supply and demand) on a new	Shippers
how long does it take for shippers to shift from one mode to the other? What are the aspects that might delay the Carriers		network?	
how long does it take for shippers to shift from one mode to the other? Shippers What are the aspects that might delay the Carriers	Infrastructure Usage	Once the new infrastructure is available,	Carriers
What are the aspects that might delay the Carriers		0 11	
			**
migration from one mode to the other? Shippers			
		migration from one mode to the other?	Shippers

implementation time and ramp-up time, shown in Table 3, and the parameter values shown in Table 4.

Finally, using the insights provided by the interviews, we were also able to estimate how the differences between stakeholders' interests could entail delays in the implementation process of the new infrastructure project and the modal shift. Fig. 1 shows three different perspectives (optimistic, realistic, pessimistic) for the planning and implementation of a new infrastructure project, using the data gained from the interviews, to show how the accumulation of different delays could entail a large difference in the time the modal shift can take place.

4. Model development

The model was built using the information retrieved from the literature review and interviews with experts. When describing the model in this chapter, we will refer to the literature sources and the interviewee results used to underpin the specific model choices.

Table 3

Implement	tation time and ramp-up time.
Expert	The slow maturation process of railways: licensing, concession, and construction takes around 10 years; The slow process allows companies to adapt to change their operation in
1	the meantime;
	There are examples of infrastructure projects where the study was
	completed at least five years ago and the project has not yet started.
	The structuring time for an infrastructure concession project in Brazil takes around 730 days;
Expert	Projects carried out by the government tend to be more time-consuming
2	due to the time required to contract studies and works;
	There is no distinction between private and public investments when it comes to implantation reliability.
	When an idea for a project comes from outside the projects already included in the government plan, usually, it is first presented (sold) to the government and this step takes years;
Expert	The regulatory/environmental aspect of the project is the most
3	bureaucratic part. It takes one year or more to get environmental licences and expropriation;
	Delays might occur during the implementation of the project. Usually, the delay in the start of the implementation of a project is greater than the delay in the implementation itself.
	In strategic planning, the delay is considered. Depending on the project, it can take three years of engineering planning only;
Expert 4	Regarding the ramp-up time, each market is different. For commodities, it goes fast as there are fewer decision-makers;
	This evolution can be fractioned too, starting with 30% of the load, then it goes to 80% then 100%.
	One of the advantages of public-private projects is that companies start to charge the user from the moment they deliver the infrastructure, an extra incentive to deliver the project on time;
Expert	For the pre-project phase, public works have a lean stage, leading to
5	greater chances of delays during the implementation due to project inconsistencies;
	The ramp-up time depends on the project. This is something specific to each project, not a systematic issue.
	Almost 100% of the grains transported by roads in Brazil have an "on- spot" set-up with truckers;
	For waterways and railways, some traders do not have a contract for their
Expert 6	transport because the logistic operating companies are subsidiaries of the trading companies themselves;
c	Because there are more alternatives for shipping cargo, traders do not feel comfortable with a long-term contract. A seasonal contract is more attractive.

4.1. Causal loop diagram

A complete visual representation of all cause-and-effect relations between the variables and feedback loops can be found in the Causal Loop Diagram (CLD), given in Fig. 2.

Decisions on new infrastructure investments rely on the available funds allocated for such projects (top right, Fig. 2). The amount of funding allocated is affected by both private and public sector investments. According to Diaz, Behr, and Ng (2016), investments in infrastructure are essential to the economy of a country, explaining the connection between investments and the country's GDP. The GDP is then linked to the volume of investments made, which is linked again to public sector investment, leading to a reinforcement loop and the government's concession agenda, which is then connected to the private sector investment reinforcement loop. Profitability expectation is an external variable that influences the private sector's decision to invest in infrastructure projects. The new infrastructure projects rely on the funds available for infrastructure-related investments in an inversely proportional relationship, as the more investment made, the less money is available in the fund, and vice versa. New infrastructure investments are also affected by geographical restrictions (such as river trafficability and environmentally protected areas) and the innovation penetration rate.

Innovation can impact possible new infrastructure investments, and therefore it should be considered when simulating the evolution of freight transportation over the up-coming decades (van Binsbergen, Konings, Tavasszy, & van Duin, 2014). In this diagram, innovations

Table 4

Parameter values retrieved from expert interviews.

Parameter	Value	Units	Source
Railway infrastructure project	10	Years	Expert 1
Ramp-up time	Instant	-	Expert 1
The possible delay between the end of the pre-project phase and the start of the implementation	5	Years	Expert 1
Structuring time for concession projects	730	Days	Expert 2
Strategic planning	15	Years	Expert 2
Environmental license and expropriation	1	Year	Expert 3
Possible delay in obtaining environmental license and expropriation	1	Year	Expert 3
Possible delay in the project implementation	5 to 8	Months	Expert 3
Trader x Operator maximum contract length	1	Year	Expert 3
Possible delay in government authorization	3 to 4	Years	Expert 4
Structure time of engineering project	2 to 3	Years	Expert 4
Increase in volume transported per year after the new infrastructure is available	30% then 80% then 100%	Tons/Year	Expert 4
Ramp-up time	3 to 5	Years	Expert 5
Material loss per transshipment	0.25%	Tons	Expert 6
Trader x Operator contract length Railway/Waterways	6	Months	Expert 6
Trader x Operator contract length Road	1	Per shipment	Expert 6
Time to reach maximum capacity	3	Years	Expert 6
Possible delay in reaching maximum capacity	2	Years	Expert 6

were aggregated into possible benefits and drawbacks that might impact the model and are represented by the green-colored variables. The innovation penetration rate is affected by both the stakeholder commitment (+) needed when researching innovation and the high implementation cost (-) required.

Infrastructure projects are split in the CLD into three main factors: infrastructure maintenance projects, increase in the network's density, and modal integration (access-point density and transshipment-related projects), all three of which influence network capacity. There are delays between new infrastructure investments and capacity because of the project's implementation time. The capacity of a certain mode further impacts the mode attractiveness, creating reinforcement loops between new infrastructure investments and mode attractiveness (Brito Junior et al., 2011; Han & Hayashi, 2008; Piattelli et al., 2002). Mode attractiveness is also influenced by generalized transport costs, which include freight tariffs, distances, travel time, and reliability. There is a balancing loop between generalized costs and freight volume, as the costs have an inversely proportional relationship with the freight volume transported.

Our CLD includes regulations to reduce CO_2 emissions that impact generalized costs, such as economic instruments, knowledge-based instruments, and new vehicle technology policies (Stelling, 2014). The implementation process of these measures is time demanding and could be affected by possible conflicts among stakeholders. After the policy measure has been chosen, it is slowly inserted into the system. These connections were made, based on the work of Gössling, Cohen, and Hares (2016).

Another key relationship in the model is between modal share, fuel consumption, CO_2 emissions, and the need for regulations (bottom of

Fig. 2). Modal share is mainly affected by mode attractiveness; the modal share is connected to the number of freights transported by each mode, which directly relates to fuel consumption. Fossil fuel consumption is proportionally linked to the amount of CO_2 emissions, which subsequently directly relates to the need to regulate these emissions. These regulations impact the generalized costs and further affect mode attractiveness in a balance feedback loop. These relationships were mainly based on Han and Hayashi's findings Han and Hayashi (2008).

4.2. SD model

Based on the CLD, we developed the stock and flow diagram, providing more detail. The main goal of the model is to analyze the modal shift over time, driven by investments made in different modes and the implementation of policy measures to reduce CO_{2e} emissions. We divided the SD model into eight modules:

- Economic-related variables;
- Freight volume generation;
- Multimodal network;
- Road saturation;
- Fiscal and regulatory policy measures;
- Vehicle-related technological innovation;
- Modal split;
- CO_{2e} emissions.

Fig. 3 presents a visual representation of all the modules that constitute the model and how they are interconnected. It also presents the main actors involved in some of them.

4.2.1. Economic-related variables

The first part of the model is the economic-related variables. In this module, we show the relationship between GDP and the volume of investment, as described in the work done by Diaz et al. (2016). The link between GDP's increase and the amount of freight transported in the network is also presented and shown in Fig. 4.

There are essential time steps between allocating funds for transportrelated investments and starting to implement the projects. These steps are part of the government's strategic planning and involve multiple aspects and stakeholders, as seen in the interviews. For this model, we aggregated the steps into three parts: (i) the pre-concession/pre-project phase, (ii) the time needed to choose which projects to implement, and (iii) the time required to obtain the environmental license and to perform land expropriation.

From Fig. 4 it can be observed that the change in GDP leads to a change in the volume of investment that both private and public sector companies can make in the network infrastructure. Therefore, the change in GDP plays a role when allocating infrastructure funds. Once the funds are allocated, it impacts each mode's capacity, as presented in the Multimodal network module.

4.2.2. Freight volume generation

The second module refers to the generation of the volume transported per mode per year. In the model, shown in figure 5, the network's freight volume is multiplied by the percentage of the modal split. By doing this, we can estimate the expected volume of freight to be transported per mode. The stocks - "Value of road volume", "Value of rail volume", and "Value of waterway volume" - accumulate the freight volume for each mode, which depend on the previous year's total freight volume and the current year's freight volume growth rate. The change in the value of roads, rails, and waterways' volume represents the increase/ decrease compared to the previous year's volume.

4.2.3. Multimodal network

The multimodal network module was initially based on Brito Junior



Fig. 1. New infrastructure project: implementation and mode shift timeline.

et al. (2011) and further adapted to the purpose of this research and the specifications of each of the available modes. For the road network (Fig. 6), the sum of funds allocated from the private and public sectors might lead to a change in road infrastructure capacity in two different ways: either by extending roads in the current network (e.g. from two lanes to four lanes) or by constructing an entirely new highway. Here, we introduce another time-related aspect to the model, as implementing a highway project takes time. Moreover, the inauguration can be done in parts, increasing the kilometers available as the construction occurs. In this model, we represent this aspect by the stocks "Amount of Km implemented" and "Amount of Km extended", which are followed by the outflows "Partially inaugurating new highway" and "Partially inaugurating extended highway", respectively. We then use these variables as inflows for "Single-lane highway network length" and "Double-lane highway network length". The passage of time and use of the infrastructure entails capacity erosion of the network, as indicated by Brito Junior et al. (2011).

With the information gathered, we can calculate the total amount of the highway network length. This critical information is used to calculate the vehicle/capacity ratio (saturation) of the road network as shown in the Road Saturation module.

For the rail network, we measure the infrastructure capacity in tons and it is influenced by the funds allocated and the infrastructure projects considered for this mode. Similar to highways, this network also loses its capacity as it is used. However, unlike the road network, rail freight in Brazil does not share its infrastructure with passenger rail. Therefore, we calculate rail transport saturation differently. On the one hand, the number of tons the network supports are converted to the maximum number of trains on the network, based on the rolling stock capacity. On the other hand, we use the rail transport demand, which is an output of the freight volume and modal split, to calculate the network's train flow. Under rail saturation, the network's installed capacity is also considered. A visual representation of the rail network and all the connections mentioned is presented in Fig. 7. The structure of the waterways network resembles the railway one. Hence, the diagram for the network is not depicted in this paper.

Converting the annual freight demand into the number of trains per day allows the establishment of a future comparison with the railway's installed capacity, which indicates the degree of saturation of the network. Once we have found the number of tons per train, it is possible to calculate the number of annual trips made by train. Because we are working on a national level, we assumed that all the trains in the network had the same characteristics. To find the number of pairs, Eq. (1) is used.

Daily number of trains = $(Tons/(Train \ capacity))/(365/(Seasonality \ index))$ (1)

To simulate the most critical scenario, in which product distribution throughout the year is not homogeneous, a seasonality index of 1.21 was used to represent the impact in the distribution due to the change of seasons, defined as the relation between 365 days within a year and the operational days of the freight (ANTT - Associação Nacional de Transportes Terrestres, 2020b). The installed capacity is defined by Eq. (2).

Installed capacity in number of trains =
$$(1, 440 x k)/(t_{cg} + \theta)$$
 (2)

where:

- The railway efficiency index linked to the maintenance of the permanent rail expressed by *k* = *k*1 *x k*2;
- 1440 = number of minutes in a day (24 h \times 60 min);
- k1 = (1,440 maintenance time)/1,440;
- *k*2 is the efficiency index linked to the management of operational resources;
- *t*_{cg} is the round-trip train time (minutes);
- θ is the licensing time (minutes).

We needed to make generalized assumptions to be able to apply this at a national level. Typically, we do this calculation for each railway yard, terminal, or both. However, we decided to use an average distance



Fig. 3. Modules included in the SD model and main actors involved.

between these points instead. We considered that the parameter values would be the same as the ones used by the government in ANTT - Associação Nacional de Transportes Terrestres (2020b), being:

- k1 = 0.9208;
- k2 = 0.8;
- $\theta = 5$ min;

The only parameter that needed to be estimated was the round-trip time (min). For this, we used the distance value of 15.85 km between each rail yard/terminal and a travel speed of 36.73 km/h (ANTT - Associação Nacional de Transportes Terrestres, 2020b).

Once we obtained the values of both the daily number of trains and installed capacity in the number of trains, the total saturation in percentage can be found using Eq. (3).

Research in Transportation Business & Management 48 (2023) 100966



Saturation (%) = (daily number of trains)/(installed capacity in number of trains)

For waterways, a similar calculation was made to obtain the necessary number of trips made by vessels to transport the demand for inland shipping. We assumed that ships could carry up to 18,000 tons each. We obtain the daily number of Vessels using Eq. (4).

Daily number of vessels =
$$(Tons/(Vessel \ capacity))/365$$
 (4)

Because there is no installed capacity in the waterway network, its saturation is measured by the ratio between the daily number of vessels and the total amount of vessels available.

4.2.4. Road saturation

As previously mentioned, road freight shares the network with passenger vehicles. We need to consider this when modeling the network's capacity, as it influences road travel time and vehicle/capacity ratio. We developed this module so that the model included this specificity, as based on Ghisolfi et al. (2019).

The first step is to add the number of road freight vehicles circulating in the network to the other vehicle categories that also use the infrastructure. The top-left part of the diagram in Fig. 8 presents steps to calculate the number of heavy vehicles. We use the freight volume and modal split to calculate the demand and divide it per vehicle capacity to find the number of heavy vehicles required to ship all the necessary goods. The bottom-left part of the diagram shows light vehicle traffic.

The steps and equations used to calculate the number of heavy and light vehicles in the network are quite similar. The main difference is that, while we calculate the heavy vehicle traffic volume using the demand and modal split percentage, we find the light vehicle traffic volume using a light-duty vehicle (LDV) growth rate that multiplies the stock variable throughout the simulation times. We assumed a constant growth of 3% per year (DNIT - Departamento Nacional de Infraestrutura de Transportes (National Department of Transportation Infrastructure), 2006).

(3)

Given the total highway traffic, it is necessary to convert it into a unit of passenger car equivalent (PCE), as defined by the Highway Capacity Manual – HCM (TRB - Transportation Research Board, 2010). To calculate road capacity, the HCM also provides the adjustment factors, due to the presence of heavy vehicles, which reduces the traffic speed of the highway. After finding out the load of the whole network, an analysis is carried out aiming to determine the relationship between equivalent traffic volume and the average capacity of the link (V/C). With the V/C ratio value, it is possible to calculate the highway travel time, using Eq. (5), provided by the Bureau of Public Roads (BPR - Bureau of Public Roads, 1964).

$$t = t_0 \left(1 + \alpha \left(\frac{V}{C} \right)^{\beta} \right)$$
(5)

where:

```
    t = average travel time;
```



Fig. 5. Freight volume generation.

- *t*₀ = free flow travel time (obtained by dividing distance by free-flow speed);
- α = calibration parameter, normally 0.15;
- $\beta = calibration$ parameter, normally 4.

Because we developed this model for a national level, we needed to estimate an average stretch distance to analyze the capacity. Furthermore, it was necessary to assess the annual average daily traffic (AADT). We also used the double-lane highway ratio in the whole highway network to assign the amount of traffic using double-lane and single-lane highways. Once we did all of these steps, we used a weighted average to find the national V/C ratio.

4.2.5. Fiscal and regulatory policy measures

Besides deciding where to allocate funds for investments, we also included other policies to promote modal shifts, such as fiscal and regulatory measures (Stelling, 2014). The measures considered include (i) fossil fuel taxation, (ii) infrastructure fee, (iii) marginal tax on CO₂, and (iv) emission trading systems (ETS). These measures can have different implementation time frames, costs, and adaptation times. We must understand the differences in the policy-making decision process and include possible delays due to stakeholders' conflicts. In the end, all these policy instruments will be converted into cost per ton and added to the generalized cost to measure the impact in the modal split. Because all the different modes have the same structure for the fiscal and regulatory measures, only the road diagram is depicted in Fig. 9.

4.2.6. Vehicle-related technological innovation

This module presents the technological innovation related to new types of vehicles. To simulate the increase in the number of new types of vehicles in the model, we needed to estimate these vehicles' penetration rate for each of the modes considered. For this model, we assumed that if the intensity of policy measures increases, so does the penetration rate of sustainable vehicles in the network, as shown in figure 10. A higher number of new vehicles in the network will have a direct impact on the amount of CO_2 produced.

4.2.7. Modal split

As initially mentioned, the generalized cost will be the main factor influencing mode choice. In this module, we applied the logit function (Ortúzar & Willumsen, 2001) to estimate the proportion of freight volume transported by each mode, considering their utility function and generalized cost. The generalized cost includes all the variables that impact the resistance to choose a mode. Such resistance is influenced by the shipment costs, the travel distance, the penalty assigned to the network because of its saturation, and the fiscal and regulatory policy measures. Moreover, for railways and waterways, the transshipment process, which includes the access/egress trip and the transshipment itself, also impacts the general cost. Eq. (6) shows how we calculated the generalized cost for roads, while Eq. (7) presents the cost calculation for rail, which is similar to the waterways cost calculation.





After calculating the modal split, the respective freight values for road, rail, and waterways feed back into the freight volume generation. We provide a visual representation of the cost calculation for roads in Fig. 11 and rail in Fig. 12. Due to its similarity to the rail cost calculation, the waterway's cost structure is not displayed.

From the shipper's perspective, it is important to look into the difference in the total costs of the modes available, based on the travel distance. Comparing the shipment costs between origin and destination allows them to choose the best alternative in terms of price. This part of the research was based on the Transport Costs Methodology, developed by the Brazilian government (EPL – Empresa de Planejamento e Logística, 2020). They estimated the transport cost for highways, railways, and waterways, considering both the fixed and variable costs. Their work also enables the estimation of each of these costs for five different commodity groups, namely: agricultural dry bulk, non-agricultural dry bulk, general freight, containerized goods, and liquid bulk. Moreover, the transshipment cost was also obtained for each of these commodity groups.

Regarding the network's saturation, we assumed that if the saturation level is below 1, the higher the saturation, the better. It allows the network to increase the trip's frequency (e.g. the number of train trips/ day), decreasing the costs needed to store commodities and the waiting time to load/unload the vehicles. The saturation only starts to become a problem once it goes above one, representing that the network is no longer able to support its demand; this leads to congestion on the roads, for example. Hence, in these cases, a penalty is added.

Another relationship that is important to highlight is the network's density. For this research, we assumed that as the infrastructure network increases, so does its density, impacting the distances needed to access a transshipment terminal. For example, after implementing new roads, the distance needed to access a barge terminal to perform the transshipment process from road to waterways might decrease. Similarly, as the rail network gets denser, the distance the trucks need to access/egress a transshipment terminal also tends to decrease. This relationship slowly decreases the road distance, and therefore the costs to access/egress such a terminal, making railways and waterways more attractive. To simplify the model, we assume that all the transhipments made in the network are road-rail-road and road-waterway-road.

4.2.8. CO_{2e} emission

 $\rm CO_{2e}$ Emission is the final module to be introduced. Here, we can estimate the yearly amount of $\rm CO_{2e}$ emission generated by freight



Fig. 8. Road saturation.



Fig. 9. Economic instruments policy.

transportation and the accumulated value over the years. To do so, we use the volume (in Tons) and the distance assigned for each mode to assess the number of Tons per mode (TKU). The calculator, provided by EPL (ONTL - Observatório Nacional de Transporte e Logística, 2021b), was used to give the value of one g CO_{2e} emitted per TKU for each mode. This module also dealt with the differentiation between emissions from sustainable vehicles and regular vehicles. It was assumed that a sustainable vehicle emits 25% less g CO_{2e} /TKU compared to standard vehicles. The ratio of each type of sustainable vehicle in the network was calculated, as shown in Fig. 13, and fed into the CO_{2e} calculation.

Fig. 14 also shows how the intensity of the policy measures is analyzed. We set up a reduction goal for CO_{2e} generated per year. If the result aligns with the goal, the policy measures do not have to be too strict. However, the higher this difference, the stricter the measures become.

Lastly, the "Accumulated CO_{2e} Emission from inter-urban freight" stock presents the accumulated volume of CO_{2e} emission during the simulation range time, as presented in Fig. 15. By having these two different stocks in the model, we can analyze how the emissions evolve yearly and how they are accumulated over time, increasing the possibility of analyzing the impact of the policy measures.

5. Verification and validation

The proposed model was verified and validated by following the SD model validation process as suggested by Forrester and Senge (1980) and Sterman (2000). The verification tests included: (i) Dimensional consistency, (ii) Time horizon, (iii) Structure verification, (iv) Extreme conditions, and (v) Integration error. The tests showed that the model was implemented accordingly to specifications. Hence, it can be compared with the structure of the system (conceptual model). The model showed consistent dimensions behaved according to the basic physical realities. The model also behaved as expected when simulated

for an 80-year long time window and also responded realistically when subjected to extreme conditions. Moreover, the model was validated by the following tests: (i) Behavior prediction, (ii) Behavior anomaly, and (iii) Behavior sensitivity, which ensured that the model is a good representation of the real system. However, it should be mentioned that assumptions were made to represent the freight environment in a closed system, meaning that this model can only be used within its boundaries.

To perform the Behavior prediction tests, a combination of three different simulations was executed. The first one is the base case scenario, where no policy measures were inserted to promote modal shifts. As an outcome, little change in the modal shift was seen, and no significant change in the volume of CO_{2e} emission per year was obtained, which is in line with the expected behavior. In the second test, the infrastructure projects were implemented in the networks, showing that the network's implementation will already entail a higher modal shift from the road to the other modes, as predicted. This also led to a decrease in the amount of CO_{2e} emission per year. In the third test, an infrastructure fee was applied to the road network on top of the implementation of new infrastructure projects, promoting an even higher modal shift and accelerating the decrease of the yearly value of CO_{2e} emission.

The Behavior anomaly test was done continuously during the development of the model. Moreover, a freight expert was involved in the final steps of the model development, checking the assumptions and validating the corresponding model behavior. Throughout the process, his remarks were evaluated and implemented, if possible. Finally, in the behavior sensitivity test, all values of the parameters that are not known were individually changed to between $\pm 10\%$ of their initial value with a random-uniform distribution. To study the results, we defined key performance indicators (KPIs) to evaluate the overall performance of the model, which are (i) the value of CO_{2e} (yearly and accumulated), (ii) the modal split, and (iii) the amount of freight volume assigned for each of the modes. For every combination of KPIs and parameters, the outcome





Per ton

Road economic

instrument cost/ton



Rail modal split

6. Applying the model - Brazilian case study

We applied the verified and validated model in the context of Brazilian freight transport. We simulated the model for 30 years, considering agricultural bulk as the commodity type because it is one of Brazil's most relevant commodities. It is also not tied to a specific mode, hence, there is competition between carriers.

instruments on road

generalised cost>

In 2020, the amount of freight volume in Brazil was 1288.5 million



Fig. 13. Ratio of sustainable vehicles in the network.

tons (ONTL - Observatório Nacional de Transporte e Logística, 2021a). Moreover, its Gross Domestic Product (GDP) was worth 1839.76 billion US dollars in 2019, according to official data from the World Bank and projections from Trading Economics (Trading Economics, 2021); this is approximately 10,056.86 billion reais (R\$). This source also provided an overview of the GDP growth rate in Brazil over the last 25 years. The historical data enabled us to analyze the country's economic stability. The lowest value achieved was -9.9% in the second quarter of 2020, and the highest was 7.7% in the following quarter. From the data, it could also be seen that the mean is approximately 0.5 and the variance is around 2. This information also enabled us to create a random distribution with the expected variation of the GDP and freight volume variation. For the base case, it was assumed that freight volume variation is directly linked to the variation of the GDP.

Other critical information is the amount of investment in infrastructure as a share of GDP. Since 2016, Brazil's investment in infrastructure has accounted for <2% of the country's GDP, having remained below 2.5% since the beginning of the decade. In 2019, almost twothirds of the total investment was made by private companies, enabling us to do a forecast with possible variations on the share of GDP invested in infrastructure projects. Furthermore, it is essential to consider that transport infrastructure is not the only type of infrastructure project in the country. So, it is also necessary to assume a percentage of the total share of infrastructure investments allocated for transport-related projects.

For the base case, we assumed that 70% of the investment in infrastructure comes from the private sector and 30% from the public sector. Moreover, within the private sector, 70% is allocated to railway projects and 30% to highway projects. While within the public sector, 80% goes to highway projects, and 20% goes to the maintenance of the waterway infrastructure.

There are essential time steps between the allocation of funds for transport-related investments and the start of the implementation of the projects. These steps are part of the government's strategic planning and involve multiple aspects and stakeholders. This process leads to an accumulation of delays. Hence, including them in the case study is crucial to understand the modal shift process due to new investments.

Other network characteristics used in the model include the value of the capacity of the trains (175,000 Tons/Train), the capacity of the vessels (18,000 Tons/Vessel), and the capacity of the trucks (57 Tons/ Vehicle). The total number of trips in the network is determined by dividing the yearly volume transported per mode by their respective capacity. Lastly, the value for the average travel distance was 630.45 km. The assumed access/egress distance from the origin to the transshipment terminal and from the transshipment terminal to the destination was 67.6 km for the railway and 113 km for the waterways.

As the research aims to analyze the promotion of modal shift as part of the decarbonization process of freight, the results assessed include the modal split and the sum of CO_{2e} emitted per mode per year. Table 5 summarizes all the scenarios analyzed in the case study.

Firstly, we propose the policy measures to be considered and apply these measures to a base case scenario to evaluate the impact they have on the system. Afterward, we introduce other four scenarios used to understand how external events might impact the system's behavior and expand the robustness of the policy measures.



Fig. 14. CO_{2e} (GHG) emissions.



Fig. 15. Accumulated CO2e (GHG) emissions.

Table	5
-------	---

Summary of scenarios analyzed in the case study.

Base case	Scenario 1	Scenario 2	Scenario 3	Scenario 4
Infrastructure investments; Fiscal and regulatory measures; Combination of both	Constant GDP growth	A higher increase in freight volume	Economic crisis	External pressure to reduce CO ₂ emissions

6.1. Base case scenario

Because we are simulating for an extended period, various projects are expected to be added to the network during the time window. The Brazilian government's plan (ANTT - Associação Nacional de

Table 6

Summary of infrastructure projects in Brazil, categorized in blocks.

2030 infrastructure block	
Activity	Km added
Highway Extension	4201.63
Highway Construction	3839.67
Railway	3007.00
2035 infrastructure block	
Activity	Km added
Highway	5471.90
Railway	4220.00
2045 infrastructure block	
Activity	Km added
Railway	3801.00

Source: Based on ANTT - Associação Nacional de Transportes Terrestres (2020a) and Investment Partnerships Program (2020).

Transportes Terrestres, 2020a) for the country's infrastructure was split into four blocks, the current infrastructure, the new infrastructure added by 2030, then 2035 and 2045, as shown in Table 6 and Fig. 16.

Based on Stelling (2014), in Table 7 we selected the following fiscal and regulatory measures: (i) Fossil fuel taxation for roads, (ii) Infrastructure fee (tolls) for roads, (iii) Marginal tax on CO_{2e} and (iv) Emission trading system. We defined a range value for each of them, as well as the respective year in which they will be implemented. We designed three-time windows for fiscal and regulatory measures, short-term (2020-2030), medium-term (2030–2040), and long-term (2040–2050). Because Brazil has no official values for marginal tax on CO_{2e} and emission trading systems, the model's values were the same as used in Europe but using the Brazilian currency. The variation in the price is based on the need to reduce CO_{2e} emissions; the higher the need, the higher the cost.



Fig. 16. Improvement of Brazil's infrastructure from 2020 to 2045. Source: ANTT - Associação Nacional de Transportes Terrestres (2020a).

Table 7

Overall fiscal and regulatory measures used in the model.

	0.0			
Fiscal and regulatory measures	Minimum value	Maximum value	Policy time window	Implementation year
Fossil fuel taxation for road	100 (R \$/m ³)	225 (R \$/m ³)	Short-term (2020-2030)	2021
Infrastructure fee (tolls) for road	8.77 (R \$/vehicle)	9.35 (R \$/vehicle)	Short-term (2020-2030)	2026
Marginal tax on CO _{2e}	38 (R \$/tCO _{2e})	50 (R \$/tCO _{2e})	Medium-term (2030–2040)	2031
Emission trading system	23.47 (R \$/tCO _{2e})	41.15 (R \$/tCO _{2e})	Long-term (2040–2050)	2040

The results of the implementation of infrastructure investments show a moderate change in the modal share over the years, with road decreasing from around 64.5% to almost 49%, while rail increased from 15.9% to nearly 24.2%, and waterways going from 19.6% to 26.8%.

The application of fiscal and regulatory measures also has had an impact on the modal shift over the years, with road decreasing from



around 64.5% to 36.8%, while waterways become the most attractive mode, increasing their market share from 19.6% to 37.6%. Meanwhile, rail increases from 15.9% to nearly 25.6%.

When combining both previous policy measures, the experiment in the base case shows that the modal shift changes more significantly, with road transportation reaching its lowest share, only 24%, while waterways become the primary mode, with 43.4% of the share, followed by railway with 32.6%. This information, combined with the shift seen in the volume transported per mode and the volume of emissions in the system, shows that the measures complement each other to promote freight decarbonization.

Some insights from this study are that rail volume benefits the most from infrastructure-related policies. In contrast, changes in waterways and road volume are more tied up to fiscal and regulatory measures than implementing the infrastructure itself. We can also see that, in an aggregated national model, fiscal and regulatory measures play a more critical role when promoting a modal shift than the need for new infrastructure. Fig. 17 shows the evolution of the modal shift over the years under the combination of the different policy measures and the respective modal split in percentage in a certain year.

Fig. 18 shows that by applying these measures we would achieve a reduction of nearly 200 $T \text{ CO}_{2e}$ in 30 years. We found that if we only implement the new infrastructures, the overall decrease in CO_{2e} in 30 years will only be 6.3%. By implementing only fiscal and regulatory



Fig. 17. Modal shift over time - combination of policy measures.



Fig. 18. Reduction of CO2e - combination of policy measures.

 Table 8

 Scenarios used to test the application of policy measures.

Scenario 1	Scenario 2	Scenario 3	Scenario 4
Constant GDP growth	A higher increase in freight volume	Economic crisis	External pressure to reduce CO ₂ emissions

Table 9

Reduction of accumulated CO2 emission.

	Accumulated CO _{2e} emission by inter-urban freight - no policy measures (gCO ₂)	Accumulated CO _{2e} emission by inter-urban freight - combination of policy strategies (gCO ₂)	Reduction (%)
Scenario 1	1.11 x 10 ¹⁵	8.46 <i>x</i> 10 ¹⁴	24.0
Scenario 2	1.28 x 10 ¹⁵	9.54 <i>x</i> 10 ¹⁴	25.4
Scenario 3	9 x 10 ¹⁴	7.27 <i>x</i> 10 ¹⁴	19.3
Scenario 4	9.87 <i>x</i> 10 ¹⁴	$6.23 x 10^{14}$	36.9

measures, the reduction is 15.5%. By combining the three measures, the reduction increases to 20.9%.

6.2. Scenarios 1-4

By applying the base case scenario, we measured the impact that different policy measures have on the decarbonization of freight transport. However, we also wanted to understand how external events might impact the system's behavior and expand the robustness of the application of such policy measures. We did this via different scenarios, which enabled us to make better recommendations about the application of policy measures. In this section, four different scenarios were designed and applied to the model, as shown in Table 8.

The assumptions we made for each of these scenarios were:

Scenario 1: Constant GDP and freight growth. Slightly stricter policy measures and a slightly higher penetration rate of new vehicles, compared to the base case scenario.

Scenario 2: Higher freight growth. Slightly stricter policy measures and a slightly higher penetration rate of new vehicles, compared to the base case scenario.

Scenario 3: Lower GDP and freight growth. Changes in policy implementation and delays on the new infrastructure projects, compared to the base case scenario.

Scenario 4: Changes in policy implementation time. A higher penetration rate of new vehicles and higher reduction of CO_{2e} from sustainable vehicles, compared to the base case scenario.

Besides the external factors defined in Table 8, we also simulated the implementation of the policies (infrastructure investment - see Table 6, and fiscal and regulatory measures - see Table 7) for each scenario. Table 9 shows the results of accumulated CO₂ emissions with no policy implemented and with all policies combined. It was observed that, despite different externalities, the modal shift still occurs, and the volume of CO₂ emissions is reduced, highlighting the efficiency of the policies simulated.

The modal shift evolution for each of the different scenarios can be seen in Fig. 19. From the base case and these four different scenarios, we can conclude that the modal shift process is slow and most of the modal



Fig. 19. Modal shift evolution over the years for the different scenarios.

Table 10

Reduction of CO2e emissions - scenario 4 extended.

	Emissions (T	rillion gCO _{2e})						
Year	2030	2040	2050	2060	2070	2080	2090	2100
Reduction (%) Reduction per decade (%)	10.0% _	28.1% 18.1%	36.9% 8.8%	41.5% 4.6%	44.9% 3.4%	47.6% 2.7%	49.6% 2%	51.1% 1.4%



Fig. 20. Extended simulation time - evolution of the emission reduction.

shift will take place between 2030 and 2040, as shown by the modal split in percentage in each of the decades. The only way of achieving significant changes in the first decade (2020-2030) is by implementing strong policy measures early on, as shown in Scenario 4.

It should be mentioned that even in the stricter scenario, the maximum level of reduced emissions reached was 36.9%, showing that much more effort is required to decarbonize the freight system. This model only looks into two different policy measures for modal shifts (infrastructure investments and a tiny sample of all the possible fiscal and regulatory options). Other decarbonization strategies, such as reducing the demand for freight, optimizing vehicle usage and loading, increasing the efficiency of freight vehicles, and promoting alternative

energy sources were left out of this research. Still, they play an essential role in the decarbonization process of freight transport. Nevertheless, these experiments shed a light on the challenge of decarbonizing the system within such a short period, and how much extra work and effort will be needed to reduce to meet even more ambitious targets, such as the EU zero emission by 2050 goal.

6.3. Extended simulation time

For this experiment, we chose the most successful scenario out of the four performed, this being the one in which we obtained the highest reduction of CO_{2e} , which was scenario 4. The following step was to

extend the simulation time to find the year when we would achieve the 50% reduction, and to check the evolution of the decarbonization rate over the following decades, as shown in Table 10.

It can be seen from Table 10 and Fig. 20 that the 50% reduction will be achieved between 2090 and 2100 (2093). In the very long-term, the decarbonization process starts to slow down, which means that the system adapts itself to the policy measures and, starts to lose its efficiency. This is valuable information for policy makers when looking for the impact of policies in the very long term and how the system needs to insert external pressure every decade or so to avoid its stabilization.

Another aspect observed in this scenario is that promoting a modal shift might not be sufficient to decarbonize freight transport. Hence, it is important to explore other alternative measures as early as possible to complement the results achieved by modal shift policies.

6.4. Discussion

Regarding new infrastructure projects, the experiments show that the increase in rail and waterways networks makes the competition tighter when choosing the mode. However, when looking at an aggregated, national level, the infrastructure project on its own does not promote a significant modal shift. We found fiscal and regulatory policies to be more effective for promoting modal shifts. The simulation shows that policies targeting the taxation of CO_{2e} emission provide a greater contribution towards making low carbon modes more attractive, compared to taxation on fuel for trucks and an infrastructure fee for highways. However, we need to keep in mind that these measures will only promote a modal shift if there are other infrastructures available for shippers to migrate to. Hence, the combination of all policy measures proves to be the most effective approach.

Interviews highlighted the bureaucratic features of implementing modal shift policies. The implementation relies on the commitments and alignments of the different stakeholders and actors involved and their interest in carrying out such a project. These aspects could lead to it taking a decade or more to implement a new infrastructure project, depending, of course, on its characteristics and complexity.

The scenario analysis proved how powerful these policy measures are, showing that decarbonization will occur under different circumstances. However, the system tends to adjust to the policy measures in the very long-term. Its impact on the overall performance tends to decrease over time. Experiments also highlighted the slow modal shift process due to the freight dynamics, which are sometimes neglected in such studies, and how this could impact the modal shift process.

Our research also shows that the SD approach allowed us to look into the complexity of the dynamics of the system and its interconnections. We studied how the variables influence each other, their non-linear cause-and-effect relationship, and the evaluation of accumulated delays and time-dependent states, analyzing how the system reacts to the policy measures. Moreover, our study looked into ways of reducing CO_{2e} emissions. For these specific policy measures, it is interesting to look at the yearly CO_{2e} emission level and its accumulated value over the years. SD enabled us to do so, providing an overview of the reduction at an accumulated level. Lastly, SD proved to be an excellent tool to measure the time aspect. Despite the benefits, SD is a method that needs some level of aggregation to be implemented, which means we have to make assumptions that could impact the reality of the system being simulated. These assumptions might lead to the loss of individual behavior, as we have to generalize carriers, shippers, and government actions. Moreover, the SD model also limits the detailed geographical differentiation, which has a significant role in studying mode choice. Lastly, the SD approach only allows equations to be used, constraining how we built some relations in the simulation model.

Regarding the generalizability of our results, we consider the direction of our main results (e.g. freight modal shift processes are slow) as being valid for other economically developing countries. The differences between this Brazilian case and other economies would mainly be related to the degree of policy inertia. The behavior of shippers and carriers, and the extent to which the political and legal system can implement new policies are important factors that will determine the speed of implementation of a modal shift and other policies and their effectiveness. These factors can differ greatly among countries.

7. Conclusions

In this paper, we explored the role of time in decarbonization policies by promoting a modal shift in freight transportation. We looked into the critical time aspects that need to be considered in this process, shedding light on the time needed to promote a shift to low carbon-intensity modes. We created an SD model that enabled us to evaluate different policy measures with different intensity levels at different times. In our research, we presented both a qualitative model (causal loop diagram) and a quantitative model (stock and flow diagram). We also used empirical data collected from interviews with experts, which gives greater validation to the model rather than using only theoretical data. The simulations showed that time issues play a crucial role in modal shift policies. In the long term, the growth of the amount of freight will impact the amount of CO2e emitted into the system. Therefore, supporting a shift to more sustainable modes relatively quickly seems vital to ensure that this growth of CO_{2e} is mitigated, despite the increase in the freight volume transported. This research showed how sustainable solutions should not be overlooked when planning for new infrastructure, and, when aligned with it, the results can be seen early on.

The simulation results showed that modal shift processes are slow, which is in line with what is observed in practice. Helping to accelerate these processes by implementing strict policy measures within the next decade could reduce CO_2 in the Brazilian context by upwards of 10% in the next thirty years, compared to the business-as-usual case. The simulation reveals that, apart from scenario 4, most of the modal shifts would take place between 2030 and 2040. Thus, its impact on the decarbonization of freight transport will mainly be seen around 10 to 20 years from now.

For future research, we recommend analyzing different fiscal and regulatory measures or even different freight decarbonization strategies, such as freight demand management, better capacity use of vehicles and assets, vehicle efficiency improvement, and the promotion of alternative fuels. Moreover, detailed behavior analysis is important to understand and reproduce shippers' and carriers' decision-making processes, which should be taken into consideration, as well as the cost-benefit analysis of the different policy measures assessed.

We also conclude that the SD modeling proved to be suitable in our case. Although it is assumption-based, we think the approach gives useful and plausible findings for policymakers about the role of time in freight CO_2 policy-making and the effectiveness of policies over time. Although the magnitude of the quantified findings should be taken with a grain of salt, the direction of the findings seems plausible and insightful. We think, therefore, that this paper contributes to knowledge, also by showing that in situations with 'messy' or incomplete data, SD can be a useful tool to help policy making by showing the decision-makers what the time dynamics of their policy proposals may be and their effectiveness in the short and long run.

Funding details

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

CRediT authorship contribution statement

Raphael Ferrari Nassar: Conceptualization, Methodology, Writing – original draft. Verônica Ghisolfi: Conceptualization, Supervision, Writing – review & editing. Jan Anne Annema: Conceptualization, Supervision, Writing – review & editing. Arjan van Binsbergen:

Conceptualization, Supervision, Writing – review & editing. Lóránt Antal Tavasszy: Conceptualization, Writing – review & editing.

Declaration of Competing Interest

The authors declare that they have no known competing financial

Appendix A. SD model equations

Table A1

Sub-model: equations for economic-related variables.

interests or personal relationships that could have appeared to influence the work reported in this paper.

Names Type of Equations Units	its
variables	
GDP Growth rate Auxiliary RANDOM NORMAL (-0.096, 0.077, 0.005, 0.02, 0) 1/Yee	/ear
Freight growth rate Auxiliary GDP Growth rate 1/Yee	
Change in Freight volume Auxiliary Freight growth rate * Freight Volume Tons Year	
Freight Volume Level Change in Freight volume Tons	
Change in GDP Auxiliary GDP * GDP Growth rate Real,	
Year	
GDPLevelChange in GDPReal% of total GDP invested in transport infrastructureAuxiliary0.005611/Ye	
% of total GDP invested in transport infrastructure Auxiliary 0.00561 1/Ye per year GDP invested in infrastructure projects Auxiliary GDP * % of total GDP invested in transport infrastructure per year Real,	
per year Year	
Change in volume of investment Auxiliary Amount of GDP invested in infrastructure projects per year * Final year for investment analysis Real,	
Final year for investment analysis Auxiliary PULSE (2020, 20) 1	ır
Final year for investment analysisAuxiliaryPULSE (2020, 20)1Volume of investmentLevelChange in volume of investment- Pre-project phaseReal	al
Pre-project phase Auxiliary Volume of investment/time needed for pre-project phase Real,	
Year	
Time needed for pre-project phaseConstant2YearInfrastructure project studyLevelPre-project phase - Choosing projectReal,	
Year	
Choosing project Auxiliary Infrastructure project study / Time needed to peak a project Real,	
Year	
Time needed to peak a project Constant 4 Year Project approval Level Choosing project - Environmental license and expropriation Real	
Project approval Level Choosing project - Environmental license and expropriation Real Environmental license and expropriation Auxiliary Project approval / Time needed to obtain environmental licences and land perform expropriation Real	
Year	
Time needed to obtain environmental licences and Constant 1 Year land perform expropriation Year	ır
Investments in new infrastructure Level Environmental license and expropriation-Yearly investment in infrastructure projects for private Real sector- Yearly investment in infrastructure project for public sector	al
Yearly investment in infrastructure projects for Auxiliary Investments in new infrastructure *(1- Ratio of total investment allocated to private sector/year) Real, public sector Year	
Yearly investment in infrastructure projects for Auxiliary Investments in new infrastructure * Ratio of total investment allocated to private sector/year Real, private sector Year	ır
Ratio of total investment allocated to private Constant 0.7 1/Ye sector/year	ear
Public sector investment in roads Auxiliary Yearly investment in infrastructure project for public sector *Percentage of yearly investment by Real, public sector put into roads Year	
Percentage of yearly investment by public sector Constant 0.8 1 put in roads	
Public sector investment in rail Auxiliary Yearly investment in infrastructure project for public sector *percentage of yearly investment by Real, public sector put into rail Year	
Percentage of yearly investment by public sector Constant 0 1 put into rail	
Public sector investment in waterways Auxiliary Percentage of yearly investment by public sector put in waterways *Yearly investment in infrastructure projects for public sector Year	
Percentage of yearly investment by public sector Constant 0.2 1 put into waterways	
Private sector investment in roads Auxiliary Percentage of yearly investment by private sector put into roads *Yearly investment in Real, infrastructure projects for private sector Year	
Percentage of yearly investment by private sector Constant 0.3 1 put into roads	1.
Private sector investment in rail Auxiliary Percentage of yearly investment by private sector put into rail *Yearly investment in Real, infrastructure projects for private sector Year	
Percentage of yearly investment by private sector Constant 0.7 1 put into rail	• •
Private sector investment in waterways Auxiliary Percentage of yearly investment by private sector put into waterways *Yearly investment in Real, infrastructure projects for private sector Year	
Percentage of yearly investment by private sector Constant 0 1 put into waterways	

Table A2

Sub-model: equations for freight demand generation.

Freight demand generation						
Names	Type of variables	Equations	Units			
Road Demand	Auxiliary	Freight Volume * Road modal split	Tons			
Change in the value of road volume	Auxiliary	(Road Demand - Value of road volume)/SAVEPER	Tons/Year			
Value of road volume	Level	Change in the value of road volume	Tons			
Road volume	Auxiliary	Value of road volume / SAVEPER	Tons/Year			
Rail Demand	Auxiliary	Freight Volume * Rail modal split	Tons			
Change in the value of rail volume	Auxiliary	(Rail Demand - Value of rail volume) / SAVEPER	Tons/Year			
Value of rail volume	Level	Change in the value of rail volume	Tons			
Rail volume	Auxiliary	Value of rail volume / SAVEPER	Tons/Year			
Waterway Demand	Auxiliary	Freight Volume * Waterway modal split	Tons			
Change in the value of waterway volume	Auxiliary	(Waterway Demand - Value of waterway volume)/SAVEPER	Tons/Year			
Value of waterway volume	Level	Change in the value of waterway volume	Tons			
Waterway volume	Auxiliary	Value of waterway volume / SAVEPER	Tons/Year			

Table A3

Sub-model: equations for the multimodal network.

Multimodal network				
Names	Type of variables	Equations	Units	
Yearly increase in railway funding Total funding for railways	Auxiliary Level	Private sector investment in rail + Public sector investment in rail Yearly increase on railway funding - Money needed to implement rail infrastructure blocks in their respective years - Yearly railway routine maintenance cost	Real/Year Real	
Yearly railway routine maintenance cost Maintenance cost /Km of rail	Auxiliary Constant	Maintenance cost /Km of rail *Rail network length 5000	Real/Year Real/Year Km	
Money needed to implement rail infrastructure blocks in their respective years	Auxiliary	(IF THEN ELSE(Total funding for railways \geq 1.04916e+11:AND: Time = 2030, 1.04916e+11, 0)) + (IF THEN ELSE(Total funding for railways \geq 1.12332e+11:AND: Time = 2035, 1.12332e+11, 0)) + (IF THEN ELSE(Total funding for railways \geq 5.3681e+10:AND: Time = 2045, 5.3681e+10, 0))	Real/Year	
Implementing new railway projects in their respective years	Auxiliary	IF THEN ELSE(Money needed to implement rail infrastructure blocks in their respective years >0:AND:Time = 2030, 3007, 0) + (IF THEN ELSE(Money needed to implement rail infrastructure blocks in their respective years >0: AND:Time = 2035, 4220, 0)) + (IF THEN ELSE (Money needed to implement rail infrastructure blocks in their respective years >0:AND:Time = 2045, 3801, 0))	Km/Year	
Change in Railway infrastructure length	Auxiliary	Implementing new railway projects in their respective years	Km/Year	
Rail network length	Level	Change in Railway infrastructure length - Rail infrastructure erosion	Km	
Rail infrastructure erosion	Auxiliary	Rail network length * Time for rail infrastructure erosion	Km/Year	
Time for rail infrastructure erosion	Constant	0.0001	1/Year	
Rail capacity improvement	Auxiliary	IF THEN ELSE(Change in Railway infrastructure length = 3007 , $6.11e+07$, 0) + IF THEN ELSE (Change in Railway infrastructure length = 4220 , $1.54134e+08$, 0) + IF THEN ELSE(Change in Railway infrastructure length = 3801 , $2.03467e+08$, 0)	Tons/Yea	
Total rail capacity increase	Level	Rail capacity improvement - Increase on rail operational capacity	Tons	
Increase on rail operational capacity Rail ramp-up time	Auxiliary	IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2031, Total rail capacity increase*0.2/SAVEPER,0) + IF THEN ELSE(Total rail capacity increase >0:AND: Time = 2032, (Total rail capacity increase*0.875/SAVEPER),0) + IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2033, (Total rail capacity increase/SAVEPER),0) + IF THEN ELSE (Total rail capacity increase>0:AND: Time = 2036, Total rail capacity increase*0.2/SAVEPER,0) + IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2037, (Total rail capacity increase*0.875/SAVEPER),0) + IF THEN ELSE(Total rail capacity increase*0.2/SAVEPER,0) + IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2038, (Total rail capacity increase>0:AND: Time = 2038, (Total rail capacity increase/SAVEPER),0) + IF THEN ELSE(Total rail capacity increase >0:AND: Time = 2046, Total rail capacity increase*0.2/SAVEPER,0) + IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2047, (Total rail capacity increase*0.875/SAVEPER),0) + IF THEN ELSE(Total rail capacity increase>0:AND: Time = 2048, (Total rail capacity increase/ SAVEPER), 0) Increase on rail operational capacity	Tons/Year	
Change in rail infrastructure capacity	Auxiliary Level	Rail ramp-up time Change in rail infrastructure capacity. Pail capacity erosion	Tons/Yea Tons	
Rail infrastructure capacity Rail capacity erosion	Auxiliary	Change in rail infrastructure capacity - Rail capacity erosion Rail infrastructure capacity * Rail transport capacity erosion rate	Tons Tons/Yea	
Rail transport capacity erosion rate	Constant	0.01	1/Year	
Train trips needed to serve total network capacity	Auxiliary	Rail infrastructure capacity / Train capacity	Train	
Train capacity	Auxiliary	Number of wagons * Capacity per wagon	Tons/Tra	
Number of wagons	Level	175	Wagons/ Train	
Capacity per wagon	Level	1000	Tons/ Wagons	
New trains trips needed to serve 100% of demand	Auxiliary	(Train trips needed to serve total network capacity / SAVEPER) - (Maximum number of trains trips in the network / SAVEPER)	Train/Yea	

(continued on next page)

Table A3 (continued)

Names	Type of variables	Equations	Units
Maximum number of trains trips in the network	Level	New trains trips needed to serve 100% of demand	Train
Maximum number of daily trains	Auxiliary	(Maximum number of trains trips in the network)/(Days per year / Seasonality index)	Train/da
Change in number of train trips/Year	Auxiliary	(Rail volume / Train capacity) - (Number of train trips / SAVEPER)	Train/Ye
Number of train trips	Level	Change in number of train trips/Year	Train/TC
-			
Daily number of train trips	Auxiliary	(Number of train trips) / (Days per year / Seasonality index)	Train/da
Seasonality index	Constant	1.21	Dmnl
Days per year	Constant	365	Days
imitation of the daily number of trains based on number of train trips available	Auxiliary	IF THEN ELSE(Daily number of train trips≥ Maximum number of daily trains, Maximum number of daily trains, Daily number of train trips)	Train/da
Rail transport saturation	Auxiliary	Limitation of the daily number of trains based on number of train trips available/Installed capacity in number of daily trains	Dmnl
nstalled capacity in number of daily trains	Auxiliary	(Minutes per day * K1 * K2)/(Round trip train time + Teta (per train))	Train/da
(1	Constant	0.9208	Dmnl
(2	Constant	0.8	Dmnl
'eta (per train)	Constant	5	Min/Tra
Ainutes per day	Constant	1440	Min/day
Round trip train time	Auxiliary	(Average round trip distance between each rail yard/terminal (per train)*2) / Rail speed * Hour to minute	Min/Tra
Iour to minute	Constant	60	Min/h
Rail speed	Constant	36.73	Km/h
Average round trip distance between each rail yard/terminal (per train)	Constant	15.85	Km/Trai
Yearly increase in waterways funding	Auxiliary	Private sector investment in waterways + Public sector investment in waterways	Real/Yea
Cotal funding for waterways	Level	Yearly increase in waterways funding - Money needed to implement waterways infrastructure blocks in their respective year - Yearly waterways routine maintenance cost	Real/Te
Zoonly waterware resisting maintenance cost	Accesilians		Deel/Ve
/early waterways routine maintenance cost Maintenance cost/km of waterway	Auxiliary Constant	Waterways network length * Maintenance cost/km of waterway 5000	Real/Yea
Monory needed to implement waterways	Accesilia	IE THEN ELSE (Total fund for unstaminants) (MND: Time $> 0, 0, 0$)	Km Real (Ver
Money needed to implement waterways infrastructure blocks in their respective year	Auxiliary	IF THEN ELSE(Total fund for waterways \geq 0:AND: Time \geq 0, 0, 0)	Real/Yea
mplementing new waterway projects in their respective years	Auxiliary	IF THEN ELSE(Money needed to implement waterways infrastructure blocks in their respective year \geq 0:AND: Time \geq 0, 0, 0)	Km/Yea
Change in waterways infrastructure length Waterways network length	Auxiliary Level	Implementing new waterway projects in their respective years Change in waterways infrastructure length - Waterway infrastructure erosion	Km/Yea Km
Waterway infrastructure erosion	Auxiliary	Waterways network length * Time for waterways infrastructure erosion	Km/Year
Fime for waterways infrastructure erosion	Constant	0.0001	1/Year
-			
Vaterway capacity improvement	Auxiliary	IF THEN ELSE(Change in waterways infrastructure length $\geq 0, 0, 0$)	Tons/Ye
Total waterway capacity increase	Level	Waterway capacity improvement - Increase on waterways operational capacity	Tons
ncrease in waterways operational capacity	Auxiliary	IF THEN ELSE(Total waterway capacity increase \geq 0:AND: Time \geq 0, 0, 0)	Tons/Ye
Vaterways ramp-up time	Auxiliary	Increase in waterways operational capacity	Tons/Ye
Change in waterways infrastructure capacity	Auxiliary	Waterways ramp-up time	Tons/Ye
Naterways infrastructure capacity	Level	Change in waterways infrastructure capacity - Waterways capacity erosion	Tons
Naterways capacity erosion	Auxiliary	Waterways infrastructure capacity * Waterways transport capacity erosion rate	Tons/Ye
Waterways transport capacity erosion rate	Constant	0.01	1/Year
Jessel trips needed to serve total network capacity	Auxiliary	Waterways infrastructure capacity / Vessel capacity	Vessel
New vessel trips needed to serve 100% of demand	Auxiliary	(Vessel trips needed to serve total network capacity / SAVEPER) - (Maximum number of vessel trips in the network / SAVEPER)	Vessel/ Year
Maximum number of vessels trips in the network	Level	New vessels trips needed to serve 100% of demand	Vessel
Maximum number of daily vessels	Auxiliary	(Maximum number of vessel trips in the network)/ (Days per year / Seasonality index)	Vessel/ days
Vessel capacity	Level	18,000	Tons/ Vessel
Change in number of vessel trips/year	Auxiliary	(Waterway volume / Vessel capacity) - (Number of vessel trips / SAVEPER)	Vessel/ Year
Number of vessel trips	Level	Change in number of vessel trips/year	Vessel
Daily number of vessel trips	Auxiliary	Number of vessel trips / (Days per year / Seasonality index)	Vessel/
(initation of the delivery the off the in- here 1	A	IE THEN ELCE/Doils number of used tring > Marine much as of deils and the	days
imitation of the daily number of trains based on	Auxiliary	IF THEN ELSE(Daily number of vessel trips \geq Maximum number of daily vessels, Maximum	Vessel/
the number of vessel trips available		number of daily vessels, Daily number of vessels trips)	days
Naterways transport saturation	Auxiliary	Limitation of the daily number of trains based on number of vessel trips available / Maximum number of daily vessels	Dmnl

Table A4

Sub-model: equations for road saturation.

Road saturation

Names	Type of variables	Equations	Units
Yearly increase in highway funding Total funding for highways	Auxiliary Level	Private sector investment in roads + Public sector investment in roads Yearly increase in highway funding - Money needed to implement highway construction infrastructure projects in their respective years - Money needed to implement highway extension infrastructure projects in their respective years - Yearly highway maintenance cost	Real/Year Real
Yearly highway maintenance cost	Auxiliary	Total single lane road routine maintenance cost + Total double lane road routine maintenance cost	Real/Year
Total single lane road routine maintenance cost Maintenance cost/Km of single lane road	Auxiliary Constant	Maintenance cost/Km of single lane road *Single lane highway network length 13.43	Real/Year Real/Year/Kn
Total double lane road routine maintenance cost Maintenance cost/km of double lane road	Auxiliary Constant	Double lane highway network length *Maintenance cost/km of double lane road 24.17	Real/Year Real/Year/Kr
Money needed to implement highway construction infrastructure projects in their respective years	Auxiliary	IF THEN ELSE(Total funding for highways \geq 3.3e+10:AND: Time = 2030, 3.3e+10,0) + IF THEN ELSE(Total funding for highways \geq 1.37302e+11:AND: Time = 2035, 1.37302e+11,0)	Real/Year
Construction of new highways	Auxiliary	(IF THEN ELSE(Money needed to implement highway construction infrastructure projects in their respective years >0:AND:Time = 2030, 3839.67, 0)) + (IF THEN ELSE(Money needed to implement highway construction infrastructure projects in their respective years >0:AND:Time = 2035, 5471.9, 0))	Km/Year
Change in Km implemented	Auxiliary	Construction of new highways	Km/Year
Amount of Km implemented	Level	Change in Km implemented - Partially inaugurating new highway	Km
Partially inaugurating new highway	Auxiliary	Amount of Km implemented/Time needed for highway to be fully opened	Km/Year
Time needed for highway to be fully opened Change in highway infrastructure length for	Constant Auxiliary	3 Partially inaugurating new highway	Year Km/Year
single lane highways Single lane highway network length	Level	Change in highway infrastructure length for single lane highways - Single lane highways	Km
		infrastructure erosion	
Single lane highway infrastructure erosion	Auxiliary	Single lane highway network length * Time for highway infrastructure erosion 0.0001	Km/Year 1/Year
Time for highway infrastructure erosion Money needed to implement highway extension infrastructure projects in their respective years	Constant Auxiliary	IF THEN ELSE(Total funding for highways \geq 5.07504e+10:AND: Time = 2030,5.07504e+10,0) + IF THEN ELSE(Total funding for highways \geq 1.425e+11:AND: Time = 2035, 1.425e+11,0)	Real/Year
Highway extension	Auxiliary	(IF THEN ELSE(Money needed to implement highway extension infrastructure projects in their respective years >0:AND:Time = 2030, 4201.63, 0)) + (IF THEN ELSE(Money needed to implement highway extension infrastructure projects in their respective years >0:AND: Time = 2035, 3007, 0))	Km/Year
Change in Km extended	Auxiliary	Highway extension	Km/Year
Number of Km extended	Level	Change in Km extended - Partially inaugurating extended highway	Km
Partially inaugurating extended highway Fime needed for highway extension to be fully	Auxiliary Constant	Number of Km extended / Time needed for highway extension to be fully delivered 3	Km/Year Year
delivered Change in highway infrastructure length for	Auxiliary	Partially inaugurating extended highway	Km/Year
double lane highways Double lane highway network length	Level	Change in highway infrastructure length for double lane highways - Double lane highways	Km
Double lane highway infrastructure erosion	Auxiliary	infrastructure erosion Double lane highway network length *Time for Highway infrastructure erosion	Km/Year
Total amount of highway network length	Auxiliary	Double lane highway network length + Single lane highway network length	Km
Vehicle capacity	Constant	57	Tons/Vehicle
increase in heavy vehicle traffic	Auxiliary	Road volume / Vehicle capacity	Vehicle/Year
Accumulated value for heavy vehicles	Level	Increase in heavy vehicle traffic	Vehicle
Accumulated value of heavy vehicles for next year Heavy vehicle flow	Auxiliary Auxiliary	Heavy vehicle traffic volume (Accumulated value of heavy vehicles for next year - Accumulated value for heavy	Vehicle Vehicle/Year
		vehicles)/SAVEPER	
Accumulated value for heavy vehicles	Level	Heavy vehicle flow	Vehicle
Heavy vehicle LDV's growth rate	Auxiliary Constant	Accumulated value for heavy vehicles / SAVEPER 0.003	Vehicle/Year Vehicle/
			Vehicle/Year
Average AADT per stretch Increase in AADT per stretch	Constant Auxiliary	22,029.6 Average AADT per stretch* LDV's growth rate	Vehicle/days Vehicle/
AADT per stretch	Level	Increase in AADT per stretch	(Year*days) Vehicle/days
Average number of vehicles/stretch/year	Auxiliary	AADT per stretch * Days per year / SAVEPER	Vehicle/Year
Average number of vehicles/year	Auxiliary	Number of highway stretch * Average number of vehicles/stretch/year	Vehicle/Year
Increase in light vehicle traffic	Auxiliary	(Average number of vehicles/year)-(Light vehicle traffic volume/SAVEPER)	Vehicle/Year
Light vehicle traffic volume	Level	Increase in light vehicle traffic	Vehicle
Accumulated value of light vehicles for next year Light vehicle flow	Auxiliary Auxiliary	Light vehicle traffic volume (Accumulated value of light vehicles for next year - Accumulated value for light vehicles) / SAVEPER	Vehicle Vehicle/Year
A	Level	Light vehicle flow	Vehicle
Accumulated value for light vehicles			
-	Auxiliary	Accumulated value for light vehicles / SAVEPER	Vehicle/Year
Light vehicles Total highway traffic	Level	Heavy vehicles + light Vehicles - Total vehicles	Vehicle
Accumulated value for light vehicles Light vehicles Total highway traffic Total vehicles Total vehicles / Highway stretch			

(continued on next page)

R.F. Nassar et al.

Table A4 (continued)

Names	Type of variables	Equations	Units
Length of highway stretch	Auxiliary	Total amount of highway network length / Average stretch distance analyzed	Dmnl
Average stretch distance analyzed	Constant	15	Km
Amount of vehicles travelling in double lane	Auxiliary	Ratio of double lane highway in total highway network* Total vehicles / Highway stretch	Vehicle/Year
network Vvhp for double lane	Auxiliary	Amount of vehicles travelling in double lane network/ (Hours per year * Number of lanes* Peak hour factor (PHF))	Vehicle/ (Hour*Lane)
Number of lanes	Constant	2	Lane
Percentage of heavy vehicles travelling in double lane network	Auxiliary	– Total heavy vehicles /Highway stretch / Amount of vehicles travelling in double lane network	Dmnl
Heavy vehicle adjustment factor (Fhv) for capacity in double lane	Auxiliary	1/((1 + percentage of heavy vehicles travelling in double lane network)*("Passenger-car equivalent for truck in a double lane network"))	Vehicle/ucp
Passenger-car equivalent for truck in a double lane network	Constant	2.5	ucp/Vehicle
Equivalent traffic volume for double lane	Auxiliary	Vvhp for double lane / Heavy vehicle adjustment factor (Fhv) for capacity in double lane	ucp/
Speed limit	Constant	80	(Hour*Lane) Km/h
Maximum capacity for double lane highways	Auxiliary	IF THEN ELSE(Speed limit≥ 100, 2200, IF THEN ELSE(Speed limit≥ 90, 2100, IF THEN ELSE(Speed limit≥ 80, 2000,1900)))	ucp/ (Hour*Lane)
Vehicle/capacity ratio for double lanes	Auxiliary	ELSE(Speed limit 2 80, 2000, 1900))) Equivalent traffic volume for double lanes/Maximum capacity for double lane highways	(Hour Lane) Dmnl
Total heavy vehicles/Highway stretch	Auxiliary	Heavy vehicle / Amount of highway stretch	Vehicle/Year
Percentage of heavy vehicles travelling in single lane network	Auxiliary	Total heavy vehicles/Highway stretch / Amount of vehicles travelling in single lane network	Vehicle/Vehic
Heavy vehicle adjustment factor (Fhv) for capacity in single lane	Auxiliary	1/(1 + Percentage of heavy vehicles travelling in single lane network*(Heavy vehicle equivalent capacity adjustment for single lane-1))	Dmnl
Heavy vehicle equivalent capacity adjustment for single lane	Constant	1.3	Dmnl
Average capacity for single lane	Auxiliary	Maximum capacity for single lane highway* "Heavy vehicle adjustment factor (Fhv) for capacity in single lane''*Grade factor for capacity adjustment	ucp/Hour
Grade factor for capacity adjustment	Constant	1	Dmnl
Maximum capacity for single lane highway	Auxiliary	Maximum capacity for single-lane highway*"Heavy vehicle adjustment factor (Fhv) for capacity in single lane"*Grade factor for capacity adjustment	ucp/Hour
Heavy vehicle adjustment factor (Fhv) for volume	Auxiliary	1/((1 + Percentage of heavy vehicles travelling in single lane network)*("Passenger-car equivalent for trucks - volume adjustment"-1))	Vehicle/ucp
Amount of vehicles travelling in single lane network	Auxiliary	Total vehicles / Highway stretch *(1-Ratio of double lane highway in total highway network)	Vehicle/Year
Vvhp for single lane	Auxiliary	Amount of vehicles travelling in single lane network/"Hours per year")/"Peak hour factor (PHF)	Vehicle/Hour
Hours per year	Constant	8760	Hour/Year
Peak hour factor (PHF)	Constant	0.9	Dmnl
Passenger-car equivalent for trucks - volume adjustment	Auxiliary	IF THEN ELSE(Vvhp for single lane≤100, 2.7, IF THEN ELSE(Vvhp for single lane≤200, 2.3, IF THEN ELSE(Vvhp for single lane≤300, 2.1, IF THEN ELSE(Vvhp for single lane≤400, 2, IF THEN ELSE(Vvhp for single lane≤500, 1.8, IF THEN ELSE(Vvhp for single lane≤600, 1.7, IF THEN ELSE(Vvhp for single lane≤700, 1.6, IF THEN ELSE(Vvhp for single lane≤800, 1.4, 1.3)))))))	ucp/Vehicle
Grade factor - volume adjustment	Auxiliary	IF THEN ELSE(Vvhp for single lane≤100, 0.67, IF THEN ELSE(Vvhp for single lane≤200, 0.75, IF THEN ELSE(Vvhp for single lane≤300, 0.83, IF THEN ELSE(Vvhp for single lane≤400, 0.9, IF THEN ELSE(Vvhp for single lane≤500, 0.95, IF THEN ELSE(Vvhp for single lane≤600, 0.97, IF THEN ELSE(Vvhp for single lane≤700, 0.98, IF THEN ELSE(Vvhp for single lane≤800, 0.99, 1)))))))	Dmnl
Equivalent traffic volume for single lane	Auxiliary	Vvhp for single lane/("Grade factor - volume adjustment"*"Heavy vehicle adjustment factor (Fhv) for volume")	ucp/Hour
Vehicle/capacity ratio for single lanes	Auxiliary	Equivalent traffic volume for single lane / Average capacity for single lane	Dmnl
Road average travel time	Auxiliary	Free flow time*(1 + Alpha*("Vehicle/capacity ratio for single lanes" <i>Beta</i>))	Hour/Vehicle
Beta	Constant	4	Dmnl
Alpha	Constant	0.15	Dmnl
Free flow time	Auxiliary	Road Distance / Free flow speed	Hour/Vehicle
Free flow speed	Constant	80	Km/h
Ratio of double lane highway in total highway network	Auxiliary	Double lane highway network length /total amount of highway network length	1
Weighted vehicle/capacity ratio	Auxiliary	((Double lane highway network length* "Vehicle/capacity ratio for double lanes") + (Single lane highway network length* "Vehicle/capacity ratio for single lanes"))/(Double lane highway network length + Single lane highway network length)	1

Table A5

Sub-model: equations for fiscal and regulatory policy measures.

Fiscal and regulatory policy measures

Fiscal and regulatory policy measures			
Names	Type of variables	Equations	Units
Starting year of fossil fuel taxation for road policy	Constant	2020	Year
Fuel taxation increase rate	Lookup	[(0,0) - (20,300)], (0,0), (1100), (2125), (3150), (4200), (5225)	Real/M ³
Fossil fuel taxation for Road	Auxiliary	STEP (Fuel taxation increase rate (Policy measures intensity), Starting year of fossil fuel taxation for road policy)	Real/M ³
Fossil fuel Cost for Road	Auxiliary	Fossil fuel taxation for Road * Truck fuel capacity	Real/Vehicle
Truck fuel capacity	Constant	0.05	M ³ /Vehicle
Fossil fuel cost added per Ton transported by road	Auxiliary	Fossil fuel Cost for Road / Vehicle capacity	Real/Tons
Starting year of road infrastructure fee	Constant	2025	Year
Number of stretches (Road)	Auxiliary	Road Distance / 630.45	Km
Road infrastructure fee increase rate	Lookup	[(0,0)-(20,10,000)],(0,0),(1,8.77),(5,9.35)	Real/Km/ Vehicle
Infrastructure fee (tolls etc.) for Road	Auxiliary	STEP ("Number of stretches (Road)"*("Road infrastructure fee increase rate"(Policy measures intensity)), Starting year of road infrastructure fee)	Real/Vehicle
Infrastructure fee per Ton transported by road	Auxiliary	Infrastructure fee (tolls etc.) for Road / Vehicle capacity	Real/Tons
Starting year of road marginal tax policy	Constant	2031	Year
Marginal Tax on CO ₂ increase rate	Lookup	[(0,0)-(20,7000)],(0,0),(1,38),(2,40),(3,50),(4,55)	Real/tCO _{2e}
Finding Marginal Tax on CO ₂ Cost	Auxiliary	STEP (Marginal Tax on CO2 increase rate (Policy measures intensity), Starting year of road marginal tax policy)	Real/tCO _{2e}
Marginal tax on CO ₂	Auxiliary	IF THEN ELSE (Emission trading systems $= 0$, Finding Marginal Tax on CO ₂ Cost, IF THEN ELSE (Finding Marginal Tax on CO ₂ Cost \geq Emission trading systems Finding Marginal Tax on CO ₂ Cost- Emission trading systems, Finding Marginal Tax on CO ₂ Cost))	Real/tCO _{2e}
Marginal tax on CO ₂ per tCO _{2e}	Auxiliary	Marginal tax on CO_2	Real/tCO _{2e}
Marginal tax on CO ₂ per gCO _{2e}	Auxiliary	Marginal tax on CO ₂ per tCO _{2e} /Converting tCO _{2e} to gCO _{2e}	Real/gCO _{2e}
Converting tCO _{2e} to gCO _{2e}	Constant	10^{6}	gCO _{2e} /tCO _{2e}
Marginal tax cost for road	Auxiliary	Road CO _{2e} emission*Marginal tax on CO ₂ per gCO _{2e}	Real/Year
Marginal cost per ton transported	Auxiliary	Marginal tax cost for road/Road volume	Real/Tons
Starting year of road ETS policy	Constant	2040	Year
Emission trade increase rate	Lookup	[(0,0) - (20,7000)], (0,0), (1,23.47), (2,39.93), (3,41.15), (4,45)	Real/tCO _{2e}
Emission trading systems	Auxiliary	STEP (Emission trade increase rate (Policy measures intensity), Starting year of road ETS policy)	Real/tCO _{2e}
Cost of ETS per tCO _{2e}	Auxiliary	Emission trading systems	Real/tCO _{2e}
Cost of ETS per gCO _{2e}	Auxiliary	Cost of ETS per tCO _{2e} /Converting tCO _{2e} to gCO_{2e}	Real/gCO _{2e}
ETS cost Road	Auxiliary Auxiliary	Road CO_{2e} emission*Cost of ETS per gCO_{2e} ETS cost Road / Road volume	Real/Year Real/Tons
ETS cost per ton transported Change in road cost per ton transported	Auxiliary	((ETS cost road volume) ((ETS cost per ton transported + Fossil fuel cost added per Ton transported by road +Infrastructure fee	Real/
change in road cost per ton transported	nuxiiuiy	per Ton transported by road + Marginal cost per ton transported by road + ministrative re- per Ton transported by road + Marginal cost per ton transported) - Impact of economic instruments on road generalized cost)/SAVEPER	(Tons*Year)
Impact of economic instruments on road generalized cost	Level	Change in road cost per ton transported	Real/Tons
ETS cost Rail	Auxiliary	Railway CO _{2e} emission*Cost of ETS per gCO _{2e}	Real/Year
ETS cost for rail per ton	Auxiliary	ETS cost Rail/Rail volume	Real/Tons
Marginal tax cost for rail	Auxiliary	Marginal tax on CO ₂ per gCO _{2e} *Railway CO _{2e} emission	Real/Year
Marginal tax for rail per train trip	Auxiliary	Marginal tax cost for rail/Rail volume	Real/Tons
Change in railway cost per ton transported	Auxiliary	((ETS cost for rail per ton + Marginal tax for rail per train trip) - Impact of economic instruments on rail generalized cost)/SAVEPER	Real/ (Tons*Year)
Impact of economic instruments on rail generalized cost	Level	Change in railway cost per ton transported	Real/Tons
ETS cost waterways	Auxiliary	Cost of ETS per gCO _{2e} *Waterways CO _{2e} emission	Real/Year
ETS cost for waterways per ton	Auxiliary	ETS cost waterways/Waterway volume	Real/Tons
Marginal tax cost for waterways	Auxiliary	Marginal tax on CO ₂ per gCO _{2e} *Waterways CO _{2e} emission	Real/Year
Marginal tax for waterways per ton	Auxiliary	Marginal tax cost for waterways/Waterway volume	Real/Tons
Change in waterway cost per ton	Auxiliary	(((ETS cost for waterways per ton + Marginal tax for waterways per ton)) - (Impact of economic	Real/
transported	T1	instruments on waterways generalized cost))/SAVEPER	(Tons*Year)
Impact of economic instruments on waterways generalized cost	Level	Change in waterway cost per ton transported	Real/Tons

Table A6

Sub-model: equations for vehicle related technological innovation.

Vehicle related technological innovation					
Names	Type of variables	Equations	Units		
Change in penetration rate of new vehicles based on policy measures	Lookup	[(0,0)-(20,300)], (0,0), (1,0.003), (2,0.00675), (3,0.0105), (4,0.015), (5,0.02)	Vehicle/ Vehicle/Year		
			(continued on next page)		

Table A6 (continued)

Vehicle related technological innovation			
Names	Type of variables	Equations	Units
Penetration rate of new road vehicle types	Auxiliary	Change in penetration rate of new vehicles based on policy measures (Policy measures intensity)	1/Year
Increase in new technology vehicles (Road)	Auxiliary	("Number of new technology vehicles (Road)"*"Penetration rate of new road vehicle types (170)") + Sustainable vehicles retired	Vehicle/Year
Number of new technology vehicles (Road)	Level	("Increase in new technology vehicles (Road)"-Sustainable vehicles retired)	Vehicle
Sustainable vehicles retired	Auxiliary	Number of new technology vehicles (Road) / sustainable vehicles lifespan	Vehicle/Year
Sustainable vehicles lifespan	Constant	15	Year
Amount of new road vehicle types	Auxiliary	Number of new technology vehicles (Road)	Vehicle
Number of sustainable vehicles in the network per day	Auxiliary	Amount of new road vehicle types / Days per year	Vehicle/days
Number of heavy vehicles in the network per day	Auxiliary	Accumulated value for heavy vehicles / Days per year	Vehicle/days
Maximum number of sustainable vehicles in the network per day	Auxiliary	IF THEN ELSE (Number of sustainable vehicles in the network per day≥ Number of heavy vehicles in the network per day, Number of heavy vehicles in the network per day, Number of sustainable vehicles in the network per day)	Vehicle/days
Change in penetration rate of new trains based on policy measures	Lookup	[(0,0) - (20,300)], (0,0), (1,0.003), (2,0.00675), (3,0.0105), (4,0.015), (5,0.02)	Train/Train/ Year
Penetration rate of new rail vehicle types	Auxiliary	Change in penetration rate of new trains based on policy measures (Policy measures intensity)	1/Year
Increase in new technology vehicles (Rail)	Auxiliary	("Number of new technology vehicles (Rail)"*"Penetration rate of new rail vehicle types") + Sustainable trains retired	Train/Year
Number of new technology vehicles (Rail)	Level	("Increase in new technology vehicles (Rail)"- Sustainable trains retired)	Train
Sustainable trains retired	Auxiliary	Number of new technology vehicles (Rail) / Sustainable train lifespan	Train/Year
Sustainable train lifespan	Constant	40	Year
Amount of new rail vehicle types	Auxiliary	Number of new technology vehicles (Rail)	Train
Number of sustainable trains in the network per day	Auxiliary	Amount of new rail vehicle types / (Days per year/Seasonality index)	Train/days
Maximum number of sustainable trains in the network per day	Auxiliary	IF THEN ELSE (Number of sustainable trains in the network per day≥ Daily number of train trips, Daily number of train trips, Number of sustainable trains in the network per day)	Train/days
Change in penetration rate of new vessels based on policy measures	Lookup	[(0,0) - (20,300)], (0,0), (1,0.003), (2,0.00675), (3,0.0105), (4,0.015), (5,0.02)	Vessel/Vessel/ Year
Penetration rate of new vessel types	Auxiliary	Change in penetration rate of new vessels based on policy measures (Policy measures intensity)	1/Year
Increase in new technology vehicles (Vessel)	Auxiliary	("Number of new technology vehicles (Waterways) * "Penetration rate of new vessel type") + Sustainable vessels retired	Vessel/Year
Number of new technology vehicles (Waterways)	Level	Increase in new technology vehicles (Vessel) - Sustainable vessels retired	Vessel
Sustainable vessels retired	Auxiliary	Number of new technology vehicles (Waterways)/Sustainable vessels lifespan	Vessel/Year
Sustainable vessels lifespan	Constant	35	Year
Amount of new waterways vehicle types	Auxiliary	Number of new technology vehicles (Waterways)	Vessel
Number of sustainable vessels in the network per day	Auxiliary	Amount of new waterways vehicle types / (Days per year/Seasonality index)	Vessel/days
Maximum number of sustainable vessels in the network per day	Auxiliary	IF THEN ELSE (Number of sustainable vessels in the network per day \geq Daily number of vessel trips, Daily number of vessel trips, Number of sustainable vessels in the network per day)	Vessel/days

Table A7

Sub-model: equations for modal split.

Aodal split			
Vames	Type of variables	Equations	Units
Road Distance	Constant	630.45	Km
Km unit	Constant	1	Km
Dimensionless conversion for total road distance cost	Auxiliary	Road Distance / Km unit	1
Road monetary value	Lookup	[(0,0) - (1500,300)], (1,15.42), (100,30.18), (200,45.08), (300,59.99), (400,74.9), (500,89.9), (600,104.71), (700,119.62), (800,134.52), (900,149.43), (1000,164.34), (1100,179.24), (1200,194.15), (1300,209.06), (1400,223.96), (1500,238.87)	Real
Total road distance cost	Auxiliary	Road monetary value (Dimensionless conversion for total road distance cost)	Real
mpact of road saturation on freight rate	Constant	0.10	Real
Saturation of road network cost	Auxiliary	Impact of road saturation on freight rate * Weighted vehicle/capacity ratio	Real
Per ton	Constant	1	Tons
Road economic instrument cost/ton	Auxiliary	Impact of economic instruments on road generalized cost*per ton	Real
Road freight transport cost	Auxiliary	((Road economic instrument cost/ton + Saturation of road network cost + Total road distance cost))	Real
Difference between road and rail distance	Constant	0	Km
Rail Distance	Auxiliary	Road Distance + Difference between road and rail distance	Km
Dimensionless conversion for total rail distance cost	Auxiliary	Rail Distance / Km unit	1

pag

Table A7 (continued)

Names	Type of variables	Equations	Unit
Rail monetary value	Lookup	[(0,0) - (1500,90)], (1,7.03), (100,12.44), (200,17.91), (300,23.37), (400,28.84), (500,34.31), (600,39.77), (700,45.24), (800,50.7), (900,56.17), (1000,61.63), (1100,67.1), (1200,72.57), (1300,78.03), (1400,83.5), (1500,88.96)	Rea
Fotal rail distance cost	Auxiliary	Rail monetary value (Dimensionless conversion for total rail distance cost)	Rea
Positive or negative impact of rail saturation	Auxiliary	IF THEN ELSE ("Rail transport saturation" ≤ 1 , —"Rail transport saturation", +"Rail transport saturation"/10)	1
mpact of railway saturation on freight rate	Constant	0.10	Rea
aturation of rail network cost	Auxiliary	Impact of railway saturation on freight rate* Positive or negative impact of rail saturation	Rea
ailway economic instrument cost/Ton	Auxiliary	Impact of economic instruments on rail generalized cost * per ton	Re
xternal costs related to rail transport	Constant	0	Re
ailway transshipment cost	Auxiliary	18.56	Re
imensionless conversion for rail network density	Auxiliary	Rail network length/Km unit	1
npact of rail network density on the amount of Km needed to be travelled by road	Lookup	[(0,0)-(80,000,60)],(0,0),(27,886.5,1e-05),(36,500,2.5),(80,000,5)	Kn
pecrease in Km needed to access/egress transhipment point for railways	Auxiliary	Impact of rail network density on the amount of Km needed to be travelled by road (Dimensionless conversion for rail network density)	Kn
nitial distance needed to access/egress transhipment point for railways	Constant	68	Kn
load transport to access/egress transhipment point for railways	Auxiliary	Initial distance needed to access/egress transhipment point for railways - ("Decrease on Km needed to access/egress transhipment point for railways")	Kn
Dimensionless conversion for road transport to access/egress transhipment point for railway	Auxiliary	Road transport to access/egress transhipment point for railways/Km unit	1
Road distance cost (access/egress) for railway	Auxiliary	Road monetary value ("Dimensionless conversion for road transport to access/egress transhipment point for railway")	Re
Fotal road transport cost to access/egress transhipment point for railway	Auxiliary	Road distance cost (access/egress) for railway *"Cost to access point x 2 (access/egress)"	Re
Cost to access point x 2 (access/egress)	Constant	2	Dr
'otal rail freight transport cost	Auxiliary	("Railway economic instrument cost/Ton" + Railway transhipment cost + Saturation of rail network cost + "Total road transport cost to access/egress transhipment point for railway" + Total rail distance cost + "External costs related to rail transport")	Re
ifference between road and waterways distance	Constant	0	Kr
Vaterways Distance	Auxiliary	Road Distance + Difference between road and waterways distance	Kr
imensionless conversion for total waterways distance cost	Auxiliary	Waterways Distance / Km unit	1
Vaterways monetary value	Lookup	$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	Re
otal waterway distance cost ositive or negative impact of waterway saturation	Auxiliary Auxiliary	Waterway monetary value (Dimensionless conversion for total waterway distance cost) IF THEN ELSE (Waterways transport saturation \leq 1, -Waterways transport saturation,	Re 1
		+Waterways transport saturation/10)	_
mpact of waterway saturation on freight rate aturation of waterway network cost	Auxiliary Auxiliary	0.10 Positive or negative impact of waterway saturation * Impact of waterway saturation on freight	Re Re
AT - to	A	rate	D.
Vaterways economic instrument cost/Ton	Auxiliary	Impact of economic instruments on waterways generalized cost*per ton	Re
external cost related to waterways transport	Constant	0	Re
Vaterways transhipment cost	Auxiliary	20.61	Re
Dimensionless conversion for road network density mpact of road network density on the amount of km	Auxiliary Lookup	Total amount of highway network length/Km unit [(0,0)-(5e+06,10)],(0,0),(109,000,1e-05),(125,400,2),(500,000,3)	1 Kı
needed to be travelled by road Decrease in Km needed to access/egress transhipment	Auxiliary	Impact of road network density on the amount of km needed to be travelled by road	Kı
point for waterways nitial distance needed to access/egress transhipment point for waterways	Constant	(Dimensionless conversion for road network density) 113	Kr
Road transport to access/egress transhipment point for waterways	Auxiliary	Initial distance needed to access/egress transhipment point for waterways - ("Decrease on Km needed to access/egress transhipment point for waterways")	Kr
Dimensionless conversion for road transport to access/egress transhipment point for waterways	Auxiliary	Road transport to access/egress transhipment point for waterways/Km unit	1
Road distance cost (access/egress) for waterways	Auxiliary	Road monetary value("Dimensionless conversion for road transport to access/egress transhipment point for waterways")	Re
Cost to access transhipment point for waterways x 2 (access/egress)	Constant	2	Dr
Total road transport cost to access/egress transhipment point for waterways	Auxiliary	Cost to access transhipment point for waterways x 2 (access/egress) *"Road distance cost (access/ egress) for waterways"	Re
Fotal waterways freight transport cost	Auxiliary	(Total waterways distance cost + "Total road transport cost to access/egress transhipment point for waterways" + "Waterways economic instrument cost/Ton" + Waterways transhipment cost + Saturation of waterways network cost + External cost related to waterways transport)	Re
Dimensionless exponent	Constant	1	Re
Road modal split	Auxiliary	(EXP(-Road freight transport cost/Dimensionless exponent) / ((EXP(-Total rail freight transport cost/Dimensionless exponent))+(EXP(-Road freight transport cost/Dimensionless exponent)) + (EXP(-Total waterways freight transport cost/Dimensionless exponent))))	Dr
Rail modal split	Auxiliary	(EXP(-Total rail freight transport cost/Dimensionless exponent) / ((EXP(-Total rail freight transport cost/Dimensionless exponent)) + (EXP(-Road freight transport cost/Dimensionless exponent)) + (EXP(-Total waterways freight transport cost/ Dimensionless exponent))))	Dn

Table A8

Sub-model: equations for CO_{2e} emissions.

CO _{2e} emissions			
Names	Type of variables	Equations	Units
Ratio of sustainable vehicles in waterway network	Auxiliary	Maximum number of sustainable vessels in the network per day / Daily number of vessel trips	1
Ratio of sustainable vehicles in road network	Auxiliary	Maximum number of sustainable vehicles in the network per day /Daily number of heavy vehicles in the network	1
Ratio of sustainable vehicles in rail network	Auxiliary	Maximum number of sustainable trains in the network per day/ Daily number of train trips	1
Amount of Tons Road	Auxiliary	Road volume * Road Distance	Km*Tons/Year
Road transportation $\mathrm{CO}_{2\mathrm{e}}$ emission rate	Constant	52	gCO _{2e} / (Tons*Km)
Road transport CO _{2e} emission rate for sustainable vehicles	Auxiliary	Road transportation CO_{2e} emission rate * 0.75	gCO _{2e} / (Tons*Km)
Road emission coming from sustainable vehicles	Auxiliary	Road transport $\mathrm{CO}_{2\mathrm{e}}$ emission rate for sustainable vehicles* Ratio of sustainable vehicles in road network	gCO _{2e} / (Tons*Km)
Road emission coming from normal vehicles	Auxiliary	Road transportation CO_{2e} emission rate * (1-Ratio of sustainable vehicles in road network)	gCO _{2e} / (Tons*Km)
Road CO _{2e} emission	Auxiliary	Amount of Tons Road*(Road emission coming from normal vehicles + Road emission coming from sustainable vehicles)	gCO _{2e} /Year
Amount of Tons Railways	Auxiliary	Rail volume * Rail Distance	Km*Tons/Year
Rail transportation CO _{2e} emission rate	Constant	18	gCO _{2e} / (Tons*Km)
Rail transportation CO _{2e} emission rate for sustainable vehicles	Auxiliary	0.75 * Rail transportation CO_{2e} emission rate	gCO _{2e} / (Tons*Km)
Rail emission coming from sustainable vehicles	Auxiliary	Rail transportation CO_{2e} emission rate for sustainable vehicles *Ratio of sustainable vehicles in rail network	gCO _{2e} / (Tons*Km)
Rail emission coming from normal vehicles	Auxiliary	Rail transportation CO_{2e} emission rate * (1-Ratio of sustainable vehicles in rail network)	gCO _{2e} / (Tons*Km)
Railway CO _{2e} emission	Auxiliary	(Rail emission coming from normal vehicles+ Rail emission coming from sustainable vehicles) * Amount of Tons Railways	gCO _{2e} /Year
Amount of Tons waterways	Auxiliary	Waterways volume * Waterways Distance	Km*Tons/Year
Waterways transportation CO_{2e} emission rate	Constant	11	gCO _{2e} / (Tons*Km)
Waterways transport C ₀₂ emission rate for sustainable vehicles	Auxiliary	Waterways transportation CO_{2e} emission rate * 0.75	gCO _{2e} / (Tons*Km)
Waterways emission coming from sustainable vehicles	Auxiliary	Ratio of sustainable vehicles in waterways network ^{$*$} Waterways transport CO ₂ emission rate for sustainable vehicles	gCO _{2e} / (Tons*Km)
Waterways emission coming from normal vehicles	Auxiliary	(1-Ratio of sustainable vehicles in waterways network) \star Waterways transportation CO _{2e} emission rate	gCO _{2e} / (Tons*Km)
Waterways CO _{2e} emission	Auxiliary	Amount of Tons waterways*(Waterways emission coming from normal vehicles + Waterways emission coming from sustainable vehicles)	gCO _{2e} /Year
Total CO_{2e} Emission by inter-urban freight	Level	(Railway CO2e emission+Road CO2e emission+ Waterways CO2e emission)-CO2e Emission per year	gCO _{2e}
CO _{2e} Emission per year Year unit	Auxiliary Constant	Total CO _{2e} Emission by inter-urban freight / SAVEPER 1	gCO _{2e} /Year Year
Dimensionless conversion to achieve goal	Auxiliary	Time/Year unit	1
CO ₂ Emission decrease rate	Lookup	[(0,0) - (3000,80)], (2020,0.999), (2030,0.895), (2040,0.737), (2050,0.55), (2060,0.363), (2070,0.176)	Dmnl
Brazilian goal for freight emission	Auxiliary	CO ₂ Emission decrease rate (Dimensionless conversion to achieve goal)	1
Reduction of CO_{2e} emissions level to achieve goal	Auxiliary	Initial value CO _{2e} *Brazilian goal for freight emission	gCO _{2e} /Year
Amount of CO ₂ emission higher than Brazilian threshold	Auxiliary	CO_{2e} Emission per year - reduction of CO_{2e} emissions level to achieve goal	gCO _{2e} /Year
gCO _{2e} unit	Constant		gCO _{2e} /Year
Dimensionless conversion for policy measures Need to regulate $C_{\rm O2}$ emission	Auxiliary Lookup	Amount of CO ₂ emission higher than Brazilian threshold/gCO _{2e} unit [(-5e+12,0) - (7.5e+13,20)], (-5e+12,1), (2e+12,1), (2e+12,1.5), (4e+12,1.5), (4e+12,2), (6e+12,2), (6e+13,3)	1 Dmnl
Policy measures intensity	Auxiliary	Need to regulate CO_2 emission (Dimensionless conversion for policy measures)	Dmnl
Accumulated CO _{2e} emission from road	Auxiliary	Road CO_{2e} emission	gCO _{2e} /Year
Accumulated CO_{2e} emission from waterways	Auxiliary	Waterways CO _{2e} emission	gCO _{2e} /Year
Accumulated CO _{2e} emission from rail	Auxiliary	Railway CO _{2e} emission	gCO _{2e} /Year
Accumulated CO_{2e} emission from inter-urban freight	Level	Accumulated CO _{2e} emission from rail + Accumulated CO _{2e} emission from road + Accumulated CO _{2e} emission from waterways	gCO _{2e}

References

- Abbas, K. A., & Bell, M. G. H. (1994). System dynamics applicability to transportation modeling. Transportation Research Part A, 28(5), 373-390.
- ANTF Associação Nacional dos Transportadores Ferroviários. (2020). Mapa Ferroviário. Available at: https://www.antf.org.br/mapa-ferroviario/ Accessed on 04 Aug. 2020.
- ANTT Associação Nacional de Transportes Terrestres. (2020a). Caderno de Demanda EF - 170 Ferrogrão: Trecho Sinop/MT - Itaituba/PA.
- ANTT Associação Nacional de Transportes Terrestres. (2020b). Caderno de estudos operacionais - EF - 170 Ferrogrão: Trecho Sinop/MT - Itaituba/PA.
- Astegiano, P., Fermi, F., & Martino, A. (2019). Investigating the impact of e-bikes on modal share and greenhouse emissions: A system dynamic approach. Transportation Research Procedia, 37, 163–170.
- Bickford, E., Holloway, T., Karambelas, A., Johnston, M., Adams, T., Janssen, M., & Moberg, C. (2014). Emissions and air quality impacts of truck-to-rail freight modal shifts in the midwestern United States. *Environmental Science & Technology, 48*(1), 446-454.

- van Binsbergen, A., Konings, R., Tavasszy, L. A., & van Duin, R. (2014). Innovations in intermodal freight transport: Lessons from Europe. In 93th annual meeting of the transportation research board, Washington (USA).
- BPR Bureau of Public Roads. (1964). Traffic assignment manual. Washington, DC, USA: Urban Planning Division, US Department of Commerce.
- Brito Junior, I., Hino, C. M., Gonçalves, P., Andrade, L. E. W. A., Moreira, C., Costa, G., ... Magalhães, D. J. (2011). Reducing CO₂ emissions due to a shift from road to cabotage transport of cargo in Brazil. *Proceedings of the International Conference of the System Dynamics Society*. 29(1). 1–27.
- Choi, B., Park, S. I., & Lee, K. D. (2019). A system dynamics model of the modal shift from road to rail: Containerization and imposition of taxes. *Journal of Advanced Transportation*, 2019, 1–8.
- Diaz, R., Behr, J. G., & Ng, M. W. (2016). Quantifying the economic and demographic impact of transportation infrastructure investments: A simulation study. *Simulation*, 92(4), 377–393.
- DNIT Departamento Nacional de Infraestrutura de Transportes (National Department of Transportation Infrastructure). (2006). Manual de Estudos de Tráfego. Instituto de Pesquisas Rodoviárias, Publicação 723. Rio de Janeiro, Brazil: DNIT.
- Elias, A. A., Cavana, R. Y., & Jackson, L. S. (2004). Analysing stakeholder dynamics in environment conflict: A new Zealand transport infrastructure project. In *The 22th* international conference of the system dynamics society, July 25–29, 2004 – Keble college, Oxford, England.
- EPL Empresa de Planejamento e Logística. (2020). Metodologia de Custos de Transporte. Available at: https://portal.epl.gov.br/manual-metodologia-de-custo s-de-transporte. Access on: 03 Dec. 2022.
- European Commission. (2011). Roadmap to a single European transport area towards a competitive and resource efficient transport system. In *White paper, communication* (p. 144).
- European Environment Agency. (2019). Trends and projections in Europe 2019 tracking progress towards Europe's climate and energy targets. In *EEA report*, 15/ 2019.
- Figgou, L., & Pavlopoulos, V. (2015). Social psychology: Research methods. International Encyclopedia of the Social and Behavioral Sciences, 22, 544–552.
- Forrester, J. W. (1961). Industrial dynamics. Waltham, MA: Pegasus Communications. Forrester, J. W., & Senge, P. M. (1980). Tests for building confidence in system dynamics
- models. In , 14. TIMS studies in the management sciences (pp. 209–228). Available at: https://www.albany.edu/faculty/gpr/PAD724/724WebArticles/ForresterSengeVa lidation.pdf. Access on 03 Dec. 2022.
- Ghisolfi, V., Ribeiro, G. M., Chaves, G. L. D., Orrico Filho, R. D., Hoffmann, I. C. S., & Perim, L. R. (2019). Evaluating impacts of overweight in road freight transportation: A case study in Brazil with system dynamics. *Sustainability*, 11(11), 3128.
- Ghisolfi, V., Tavasszy, L. A., Correia, G. H. D. A., Chaves, G. D. L. D., & Ribeiro, G. M. (2022). Freight transport decarbonization: A systematic literature review of system dynamics models. *Sustainability*, 14(6), 3625.
- Gössling, S., Cohen, S. A., & Hares, A. (2016). Inside the black box: Eu policy officers' perspectives on transport and climate change mitigation. *Journal of Transport Geography*, 57, 83–93.
- Han, J., & Hayashi, Y. (2008). CO₂ mitigation scenarios in China's inter-city freight transport. Civil Engineering Planning Research Presentation, 37(1), 1–4.
- Heinen, A. (2021). The impacts of e-commerce growth on the amount of kilometers driven for last-mile delivery in the Netherlands: A system dynamic based analysis on the relation between e-commerce and the logistic sector. In *Master thesis (transport, infrastructure and logistics programme)*. Delft University of Technology. Available at: http://resolver.tudelft.nl/uuid:e5e82825-3e89-430a-af24-2a011914b737. Access on 03 Dec. 2022.
- Hou, L., & Geerlings, H. (2016). Dynamics in sustainable port and hinterland operations: A conceptual framework and simulation of sustainability measures and their effectiveness, based on an application to the port of Shanghai. *Journal of Cleaner Production*, 135, 449–456.

- Investment Partnerships Program. (2020). PPI in numbers 2019-2022. Available at: https://www.ppi.gov.br/ppi-english Accessed on: 30 Jul. 2020.
- Jonkeren, O., Francke, J., & Visser, J. (2019). A shift-share based tool for assessing the contribution of a modal shift to the decarbonization of inland freight transport. *European Transport Research Review*, 11(1), 1–15.
- Kaack, L. H., Vaishnav, P., Morgan, M. G., Azevedo, I. L., & Rai, S. (2018). Decarbonizing intraregional freight systems with a focus on modal shift. *Environmental Research Letters*, 13(8), Article 083001.
- Liu, P., Mu, D., & Gong, D. (2017). Eliminating overload trucking via a modal shift to achieve intercity freight sustainability: A system dynamics approach. *Sustainability*, 9 (3), 398.
- Nepveu, M. R. (2020). Implementing urban waterway transport as a sustainable freight transport solution: A case study for the city of Amsterdam. In *Master thesis (complex* systems engineering and management (CoSEM) programme). Delft University of Technology. Available at: http://resolver.tudelft.nl/uuid:bb6a74ab-81e3-4cb 8-808f-a87617ca9a2f. Access on 03 Dec. 2022.
- ONTL Observatório Nacional de Transporte e Logística. (2021a). Database. Available at: https://ontl.epl.gov.br/exploredados/consultasabasededados/ Accessed on: 15 Feb. 2021.
- ONTL Observatório Nacional de Transporte e Logística. (2021b). Simulador de Gases de Efeito Estufa (GEE). Available at: https://ontl.epl.gov.br/aplicacoes/simulador-de -gases-de-efeito-estufa-gee/ Accessed on: 10 Mar. 2021.
- Ortúzar, J. D., & Willumsen, L. G. (2001). Modeling transport (3rd ed.). New York, NY, USA: JohnWiley & Sons (ISBN 9780471861102.).
- Piattelli, M. L., Cuneo, M. A., Bianchi, N. P., & Soncin, G. (2002). The control of goods transportation growth by modal share re-planning: The role of a carbon tax. System Dynamics Review, 18(1), 47–69.
- Pruyt, E. (2013). Small system dynamics models for big issues: Triple jump towards real-world complexity (p. 324). Delft: TU Delft Library.
- Purwanto, J., González, I. H., Vanherle, K., Fermi, F., & Fiorello, D. (2011). Global scale system dynamic simulation model for transport emissions (GLADYSTE). Transporti e Territorio – TRT. Available at: http://www.trt.it/documenti/GLADYSTE.pdf Accessed on: 01 Jul. 2021.
- Saryazdi, A. H. G., Ghatarim, A. R., Mashayekhi, A. N., & Hassanzadeh, A. (2020). Group model building: A systematic review of the literature. *Journal of Business School*, 3(3), 98–136. https://doi.org/10.26677/TR1010.2021.631
- Seitz, C. (2014). Conceptual causal framework for the diffusion of emerging CO2-saving technologies in heavy commercial vehicles. *Karlsruher Institute of Technology*, 25, 1–24.
- Shepherd, S. P. (2014). A review of system dynamics models applied in transportation. *Transportmetrica B: Transport Dynamics*, 2(2), 83–105.
- Stelling, P. (2014). Policy instruments for reducing CO₂-emissions from the Swedish freight transport sector. *Research in Transportation Business & Management*, 12(1), 47–54.
- Sterman, J. (2000). Business dynamics: System thinking and modeling for a complex world (1st ed.). New York, NY, USA: McGraw-Hill. ISBN 978–0072389159.
- Thaller, C., Clausen, U., & Kampmann, R. (2016). System dynamics based, microscopic freight transport simulation for urban areas. In U. Clausen, H. Friedrich, C. Thaller, & C. Geiger (Eds.), *Commercial transport. Lecture notes in logistics*. Cham: Springer. https://doi.org/10.1007/978-3-319-21266-1_4.
- Trading Economics. (2021). Brazil GDP. Available at: https://tradingeconomics.com/br azil/gdp Accessed on: 15 Feb. 2021.
- TRB Transportation Research Board. (2010). Highway capacity manual (p. 2010). National Research Council: Washington, DC, USA. ISBN 978-0-309-16077-3.
- Woodburn, A., Browne, M., Piotrowska, M., & Allen, J. (2007). Litterature review WM7: Scope for modal shift through fiscal, regulatory and organisational change. Transport Studies Group, University of Westminster and University of Leeds.